



# Article Neuro-Advancements and the Role of Nurses as Stated in Academic Literature and Canadian Newspapers

Rochelle Deloria<sup>1</sup> and Gregor Wolbring<sup>2,\*</sup>

- <sup>1</sup> Cumming School of Medicine, University of Calgary, Calgary, AB T2N4N1, Canada
- <sup>2</sup> Community Rehabilitation and Disability Studies, Department of Community Health Sciences, Cumming School of Medicine, University of Calgary, Calgary, AB T2N4N1, Canada
- \* Correspondence: gwolbrin@ucalgary.ca

Received: 14 April 2019; Accepted: 22 August 2019; Published: 26 August 2019



Abstract: Neurosciences and neurotechnologies (from now on called neuro-advancements) constantly evolve and influence all facets of society. Neuroethics and neuro-governance discourses focus on the impact of neuro-advancements on individuals and society, and stakeholder involvement is identified as an important aspect of being able to deal with such an impact. Nurses engage with neuro-advancements within their occupation, including neuro-linked assistive technologies, such as brain-computer interfaces, cochlear implants, and virtual reality. The role of nurses is multifaceted and includes being providers of clinical and other health services, educators, advocates for their field and their clients, including disabled people, researchers, and influencers of policy discourses. Nurses have a stake in how neuro-advancements are governed, therefore, being influencers of neuroethics and neuro-governance discourses should be one of these roles. Lifelong learning and professional development could be one mechanism to increase the knowledge of nurses about ethical, social, and legal issues linked to neuro-advancements, which in turn, would allow nurses to provide meaningful input towards neuro-advancement discussions. Disabled people are often the recipients of neuro-advancements and are clients of nurses, therefore, they have a stake in the way nurses interact with neuro-advancements and influence the sociotechnical context of neuro-advancements, which include neuro-linked assistive devices. We performed a scoping review to investigate the role of narrative around nurses in relation to neuro-advancements within academic literature and newspapers. We found minimal engagement with the role of nurses outside of clinical services. No article raised the issue of nurses having to be involved in neuro-ethics and neuro-governance discussions or how lifelong learning could be used to gain that competency. Few articles used the term assistive technology or assistive device and no article covered the engagement of nurses with disabled people within a socio-technical context. We submit that the role narrative falls short of what is expected from nurses and shows shortcomings at the intersection of nurses, socio-technical approaches to neuro-assistive technologies and other neuro-advancements and people with disabilities. Neuro-governance and neuroethic discourses could be a useful way for nurses and disabled people to co-shape the socio-technical context of neuro-advancements, including neuro-assistive technologies. Lifelong learning initiatives should be put in place to provide the knowledge necessary for nurses to take part in the neuroethics and neuro-governance discussion.

**Keywords:** neuro-advancement; neuroethics; neuro-governance; nurses; nursing; governance; engagement; lifelong learning; role; newspapers; scoping review; academic literature

#### 1. Introduction

Neuroscientific and neurotechnological advancements (from now on called neuro-advancements) have a long history and continue to develop and grow [1–7]. Neuroscience is "… the study of how the nervous system develops, its structure, and what it does" [8] and neurotechnologies are defined for example, "… as the assembly of methods and instruments that enable a direct connection of technical components with the nervous system" [9], p. 1.

Applications of neuro-advancements range from therapeutic to recreational, entertainment, military, human enhancement and education. One can classify various neuro-advancements as neuro-linked assistive technologies, such as brain-computer interfaces (BCI), cochlear implants, and virtual reality.

In response to the broad applicability of neuro-advancements, neuroethics and neuro-governance discourses arose to identify potential ethical, social, and legal issues that neuro-advancements raise. Stakeholder involvement is one focus of neuroethics and neuro-governance discourses.

Health professionals in general are stakeholders of neuro-advancement discourses, as they interact with both therapeutic and non-therapeutic neuro-advancements within their occupation and daily lives [10,11]. Nursing acknowledges this interaction by having generated the specialty of neuroscience nursing [12–15].

Within their occupation, nurses uphold multiple roles, such as educator, advocate, influencer, researcher, and clinical care provider [16–18].

However, an article by Bell, Legar, Sankar, and Racine indicated gaps within the current roles of health professionals, including nurses, in relation to neuro-advancements, stating:

"Trainees from diverse healthcare professions (e.g., nursing, social work, physiotherapy) are not well prepared to handle many of the ethical issues associated with psychiatric DBS [deep brain stimulation] because, among other reasons, they may be unprepared to engage in ethical reflection, they have a limited understanding of issues associated with scientific uncertainty, and they may lack an interdisciplinary understanding about ethical issues". [19], p. 6

Another group that has a stake in how neuro-advancements, such as the socio-technical context of neuro-linked assistive devices, are discussed, are disabled people. Nurses often use neuro-advancements for assisting their clients, which include disabled people. Thus, the way in which nurses use, perceive and interact with various neuro-advancements is one factor that affects how neuro-advancements are discussed by nurses in relation to disabled people.

Given the role expectations of nurses, their interactions with clients, such as disabled people, and the gaps in ethical knowledge in health professionals identified [19], our objective was to better understand the role expectations of nurses evident in the academic literature and newspaper literature surrounding neuro-advancement, and whether the literature engaged with the topic of nurses obtaining the necessary knowledge in order to contribute effectively to the ethical and governance discussions of neuro-advancements. Although the above rationale for this study applies to other health professionals as well, we chose nursing for this scoping review because initial research indicated that nursing generated a greater amount of data we could analyze than other health professions.

#### 1.1. Scope of Neuro-Advancements

The scope of neuro-advancements ranges from medical/therapeutic, military, education, human enhancement to recreational, including the use of mainstream neuro-assistive technologies by disabled people and mainstream society [10,11,20–24]. Some products and enabled procedures are transcranial direct stimulation techniques, neurosurgery, deep brain stimulation, artificial hippocampus, artificial brain, prosthetic memory device, cognitive enhancement and nootropic drugs [22,25–28]. As the innovation of neuro-advancements continues to grow, it has begun to converge outside the clinical and research settings, blurring the line between therapeutic and non-therapeutic use [10,11,29,30]. With the rise of "dual use" technologies of " ... any/all applications beyond clinical care, to include

lifestyle optimization, military/warfare and political/security operations" [11], p. 12, and DIY neuroadvancements [30], international groups such as the *European Union Human Brain Project* [31] and the *Nuffield Council on Bioethics* [7] have started to shape the conversation about the use of these applications within mainstream society.

# 1.2. Neuroethics and Neuro-Governance

The societal discussions of science and technology advancements should be a constant endeavor [32,33]. Various terms have emerged to discuss how to govern science and technology, such as democratizing science and technology [34–38], participatory technology assessment [39–43], anticipatory governance [44–46], upstream engagement [47–50] and responsible innovation [51–55]. Neuroethics was coined as a term and further developed as a field to investigate various ethical, social, and legal issues raised by neuro-advancements as a whole [56–63] and in relation to particular neuro-advancements, such as brain computer interfaces (BCI) [64,65] and deep brain stimulation (DBS) [19]. A 2018 Organization for Economic Co-operation and Development (OECD) report on the issues of neurotechnology governance argued for the importance of understanding the ethical effects of neuroscience and that neurotechnology governances are necessary to ensure responsible advancements and innovation within the neuro field [11]. The same report said:

"Trust and trust-worthiness, including the development of transparent governance with inclusive participation, are important factors in ensuring public support for new approaches in science and technology and will rightly impact professional and public acceptance". [11], p. 5

According to Racine, "... neuroethics has given considerable attention to topics related to public engagement, public understanding of neuroscience, and the media" [66], p. 1467. The use of media in distributing information can be a key component in the widespread release of information about neuro-advancements towards the public eye [66,67].

# 1.3. Role and Skills of Nurses

"Nursing is a practice discipline and it is a political act" [17], p. 1. According to the Canadian Nurses Association, the skillset of nurses ranges from providing appropriate access to health care and safe guarding human rights [68] to reading, critiquing and applying evidence-informed literature and participating in research [69]. To ensure high-quality nursing practice, nurses must apply educational, clinical, professional, ethical and social expertise through a supportive and consultative role in order to meet the needs of their patients and advocate for policy change to develop the nursing profession as an entirety [17,70]. The Canadian Nurses Association outlined eight essential nursing competencies for 2020:

- 1. "A global perspective or mindset regarding health care and professional nursing issues;
- 2. A working knowledge of technology to facilitate the mobility and portability of relationships, interactions, and operational processes;
- 3. Expert decision-making skills rooted in empirical and evidence-based science and research;
- 4. prioritizing quality and safety;
- 5. Being politically astute, by understanding and appropriately intervening in political processes;
- 6. Collaborative and team building skills
- 7. Balancing authenticity and performance expectations
- 8. Coping effectively with change by proactively envisioning and adapting to a constantly variant healthcare system characterized by rapid change and chaos" [17].

Similarly, the American Nurses Association (ANA) and the American Association of Colleges of Nursing (AACN) outlined the following responsibilities that all nurses should encompass, regardless of their specialty [18]:

- 1. "1. Plans and coordinates interventions from a multidisciplinary perspective.
- 2. Functions across the healthcare system and works with diverse populations.
- 3. Initiates and facilitates quality improvement initiatives.
- 4. Facilitates, conducts, and promotes utilization of research activities in practice.
- 5. Develops education strategies and evaluates effectiveness of educational interventions.
- 6. Recommends and influences social and healthcare policies.
- 7. Provides consultation to improve care.
- 8. Applies legal and ethical standards to complex situations."

The American Nurses Association (ANA), the American Association of Colleges of Nursing (AACN) and the Canadian Nurses Association (CNA) acknowledged that all nurses within the workforce must communicate and engage themselves at the matters of their education, and the matters of social and ethical affairs aside from their clinical obligations. With this skillset, nurses execute the role of employee/worker, advocate, medical professional, educator, and leader/expert. The proper utilization and development of these skillsets and knowledge is prompted through lifelong learning and educational initiatives [71].

# 1.4. Lifelong Learning and Nurses

"Learning is more process-oriented than product-oriented. More learning needs more time. The total amount of time devoted to learning over a lifespan, as well as the amount of time spent in learning activities parallel to everyday life, is increasing". [72]

Many voiced the importance of lifelong learning including the European Union (EU), the Organization of Economic Cooperation and Development (OECD) and The United Nations Educational, Scientific, and Cultural Organization (UNESCO). The OECD project, "Brain and Learning," indicated the importance of lifelong learning and the implementation of neuro background [73,74].

"Not only must individuals acquire initial training as advanced as possible in order to enhance their performance and increase their chances of entering the labour market, but once they have entered the market, they must embark upon a process of lifelong learning to preserve their position in a fiercely competitive environment. The development of the "learning society" further increases the need for individual learning". [74], p. 2

The continuation of nursing education through professional development initiatives allows for the spread of information by nurses, facilities, agencies, and institutions to help enhance nursing competencies [17,70,75]. The proper engagement of the individual nurse towards lifelong learning is reflected in " ... quality of nursing practice" [75], p. 4. through updated patient care, awareness of new innovations and technologies, development and acquiring skillsets, and obtaining the relevant knowledge needed for change and advocacy [17,70,75,76]. Lifelong learning was also acknowledged as important for neuroscience nursing [77].

In our study, we investigated further the collaboration between lifelong learning initiatives and the exposure of neuro-advancements within the nursing field.

# 1.5. People with Disabilities and Neuro-Advancements

The importance of assistive technologies for disabled people has been indicated in the United Nations' Convention on the Rights of Persons with Disabilities [78] and various neuro-technologies could be classified as neuro-assistive technologies or neuro-assistive devices (e.g., cochlear implants, brain–computer interface, virtual reality) [7,79–82]. Due to this relationship, the narrative surrounding neuro-advancements has an effect on disabled people. Numerous problems have been identified in how assistive technologies are discussed, such as that the very definition of assistive technology being seen as too medical [83]. Therefore, it is important to analyze the narrative that exists for

neuro-assistive devices and neuro-advancements. Indeed, it has already been critiqued that brain computer interfaces are discussed purely within the framework of disabled people as therapeutic users of the neuro-technology and not as non-medical user of brain computer interfaces, which is reserved for non-disabled people [81].

There are many groups that interact with neuro-advancements, such as engineers and various allied health professions who often have disabled people as clients that are envisioned to be users of neuro-advancements. Nurses are one occupation covered under allied health professionals that interact with neuro-advancements and are often the first line of contact to analyze how disabled people interact with and respond to the use of diverse types of neuro-advancements. At the same time, the role expectation for nurses [17,70] includes being advocates for their client and their field and the responsibility to influence social policies [18]. These roles suggest that nurses should take part in neuroethics and neuro-governance discussions. Lifelong learning, an action expected from nurses, has the potential to be one mechanism to allow individuals to gain knowledge on neuro-advancements, including the ethical, social, legal and other non-clinical aspects. This would allow nurses to exercise a voice in increasing the benefits of neuro-advancements, to decrease the dangers of neuro-advancements for their clients, such as disabled people, and to be able to influence the sociotechnical context around neuro-advancements, including neuro-assistive technologies.

To conclude, the background section shows that nurses have a stake in neuro-advancements, including how neuro-advancements are governed. Nurses use various neuro-advancements as part of their professional practice and clients, such as disabled people, use neuro-advancements such as neuro-linked assistive technologies. As such, nurses should contribute to the socio-technical context discussions around assistive technologies and disabled people. Nurses are expected to perform many roles and these roles could be employed to engage with neuro-advancements beyond clinical service provision, such as influencing neuroethics and neuro-governance discourses. However, it has been reported that nurses lack knowledge in relation to neuroethics issues. Lifelong learning is the vehicle used by nurses to stay up to date on topics and could be used to obtain the reported missing knowledge.

# 2. Materials and Methods

#### 2.1. Study Design

#### Academic and Newspapers

A scoping review approach was the most suitable for our study. One purpose of scoping studies is to identify the extent of research present on a given topic [84–89] and the current understanding of a given topic. As proposed by Arksey and O'Malley [87], we utilized the recommended five step process of scoping studies: identifying the review's research purpose and question, identifying databases to search, recording the descriptive quantitative results, selecting literature based on descriptive quantitative results for qualitative analysis, qualitative analysis of data, and reporting findings of qualitative analysis. We did not perform the optional step [6] of consulting stakeholders to execute knowledge translation for this study because that is part of a follow up study.

#### 2.2. Identifying and Clarifying the Purpose and Research Questions

#### Academic and Newspapers

Our objective was to understand the role narrative of nurses in neuro-advancements covered by academic literature and newspapers, including the role of nurses in relation to disabled people using neuro-advancements and whether the literature engaged with the essential topic of nurses obtaining the necessary knowledge to engage in and contribute to discussions around all aspects of neuro-advancements, including neuroethics and neuro-governance discussions and the socio-technical context of neuro-linked assistive technologies.

To fulfill our objective, we first asked the following question:

How often are nurses and nursing mentioned in conjunction with the terms "assistive technology\*" or "assistive device\*" in academic literature and various neuro-terms in academic literature and newspapers? This question allowed us to identify data to be downloaded for the content analysis.

Our content analysis of the obtained academic abstracts and Canadian newspaper articles was guided by the following questions:

Which roles and skills of nurses are mentioned in relation to neuro-advancements? How is lifelong learning in relation to neuro-advancements and nurses engaged with? Which ethics issues are identified in relation to neuro-advancements? How are "assistive technology" and disabled people mentioned in relation to neuro-advancements? Which governance issues are mentioned in relation to neuro-advancements, if any?

#### 2.3. Data Sources and Data Collection

#### 2.3.1. Newspapers

It has been well reported that newspapers and other media influence discourses [90]. The media has a significant impact on individuals and is influential in creating their perspective on many issues. It sets an opportunity for flow of information and new perspective [91]. To conduct our newspaper search, we used the *Canadian Newsstream*, a database consisting of n = 300 English Language Canadian newspapers, with a complete time range from 1980–2018. We collected articles from these newspapers using the University of Calgary's ProQuest online database.

Canadian newspapers were chosen because (a) existing and potential nurses are readers of newspapers, (b) the database covers over 300 news sources covering all regions of Canada, (c) over 75% of Canadians still read newspapers [92,93] and as such, are influenced by what they read and (d) Parents, teachers, and career counselors who often give advice on career ideas to young adults are also readers of newspapers [94].

#### Eligibility Criteria and Search Strategies

To maintain a clear and feasible scope [89], eligible newspaper articles were identified using explicit search strategies [85]. On 28 March 2018 we employed two search strategies: first, we searched the newspaper database for the term "nurs\*" in combination with "neurote\*" OR "neurosc\*" or "neuroen\*", obtaining n = 700 results after n = 84 duplicates were removed. Secondly, we searched the database for "nurs\*" in conjunction with the select 70 neuro-advancement terms (Table 1) obtaining n = 695 hits after n = 12 duplicates were removed. We downloaded the n = 1365 newspaper articles we found as two individual PDF, reflecting the two newspaper search strategies and imported the PDF into ATLAS.Ti 8<sup>TM</sup>, a qualitative data analysis software application for qualitative content analysis.

The following 70 neuro-advancement-related terms, which we identified during our prior research, were used for both academic literature and newspapers:

#### 2.3.2. Academic Literature

We searched two academic databases, namely EBSCO ALL, an umbrella database that includes over 70 other databases, itself including the nursing focused database CINAHL and Scopus, which incorporates the full Medline database collection with no time restrictions. We chose these two databases because together, they covered a wide range of topics from areas of relevance to answer the research questions, such as life science, physical science, nursing, neuroscience, disability studies, social science and technology and relevant journals, such as the journal called *Journal of Neuroscience Nursing*.

1.	artificial brain	36.	HEG
2.	artificial hippocampus	37.	hemoencephalogra*
3.	auditory brainstem	38.	Hippocampus prosthesis
4.	BCI	39.	human computer
5.	bionic eye	40.	iEEG
6.	brain computer	41.	intracranial electroencephalography
7.	brain feedback*	42.	Muse headband
8.	brain imaging	43.	Neural stem cell
9.	brain inform*	44.	Neuralink
10.	brain stimul*	45.	Neurochip
11.	brain to speech	46.	neurofeedback*
12.	Brain-computer interface	47.	neuroimaging
13.	brain-to-text	48.	neuroinform*
14.	cochlear implant	49.	neuromodula*
15.	cognitive enhance*	50.	neurosens*
16.	cognitive imag*	51.	neurostim*
17.	cognitive stim*	52.	NFB
18.	cognitive tech*	53.	Nootropic
19.	collaborative cognitive simulations	54.	NSC
20.	CoriQ	55.	optogenetic
21.	cortical modem	56.	Prosthetic memory device
22.	cranial electrotherapy stim*	57.	Pulsed electromagnetic field therapy
23.	DACI	58.	Responsive neurostimulation
24.	darpa ram sensor	59.	Sacral nerve stimulation
25.	DBS	60.	Speech BCI
26.	deep brain stim*	61.	Spinal cord stimulator
27.	deep mind	62.	subvocal
28.	Ear-EEG	63.	tDCS
29.	ECoG	64.	THync mood altering headset
30.	EEG biofeedback	65.	TMS
31.	Electrocorticogra*	66.	transcranial direct current stim*
32.	Exocortex	67.	transcranial magnetic stim*
33.	Facial electromyography	68.	virtual reality
34.	fEMG	69.	VR

 Table 1. List of neuro-advancement search terms.

Eligibility Criteria and Search Strategies

In EBSCO, we only searched for scholarly peer reviewed journals, while in Scopus, we searched for reviews, peer reviewed articles, conference papers, and editorials.

We applied two search strategies on 28 March 2019:

Strategy 1a:

We searched the abstracts of articles in EBSCO ALL and Scopus for the term "nurs\*" in combination with "neurote\*" OR "neurosc\*" or "neuroen\*".

Strategy 1b:

We searched the title of journals in Scopus for the term "nurs\*" in combination with "neurote\*" OR "neurosc\*" or "neuroen\*" in the abstracts.

In total, we obtained n = 1522 results after we removed n = 1655 duplicates.

Strategy 2a:

We searched the abstracts of articles in EBSCO ALL and Scopus for the term "nurs\*" in conjunction with the select 70 neuro-advancement terms (Table 1).

Strategy 2b:

We searched the titles of journals in Scopus for the term "nurs\*" in conjunction with the select 70 neuro-advancement terms (Table 1) in the abstracts.

In total, we obtained n = 597 abstracts after elimination of n = 123 duplicates.

On 1 July 2019, we performed an additional search.

#### Strategy 3a:

We searched the abstract of articles in EBSCO ALL and Scopus for the term "nurs\*" in conjunction with the terms "assistive technolog\*" or "assistive device\*".

#### Strategy 3b:

We searched the titles of journals in Scopus for the term "nurs\*" in conjunction with the conjunction with the terms "assistive technolog\*" or "assistive device\*" in abstracts.

In total, we obtained n = 344 abstracts after elimination of n = 193 duplicates.

We exported all abstracts into a word document and the word document was imported into ATLAS.Ti 8<sup>TM</sup>, a qualitative data analysis software application for content analysis.

# 2.4. Quantitative Data Analysis and Qualitative Data Reporting

# Newspapers and Academic Newspapers

We employed a descriptive quantitative (manifest coding) and qualitative content analysis (latent coding) to answer our research questions.

To answer the research question of which neuro-advancements are present in the covered literature, both authors performed manifest coding that involved obtaining the "frequency and location of a certain "recording unit" [75], p. 47 for the number of abstracts downloaded that contained any of the 70 neuro-related terms (Table 1).

Both authors independently performed latent coding of the downloaded academic abstracts and full-text newspaper articles to discover "underlying meanings of the words or the content" [78], p. 1284. Both authors idependently grouped related codes for content relevant to the scope of the study, namely (1) what roles of nurses are mentioned in conjunction with neuro-advancement? (2) How are disabled people mentioned in relation to nurses/nursing and neuro-advancements? (3) What ethics issues are mentioned in relation to nurses/nursing and neuro-advancements? (4) Was the role of nurses being involved in neuroethics and neuro-governance discussions present? (5) How are assistive technologies/assistive devices mentioned in relation to neuro-advancements and nurses/nursing?

Much of the content was not relevant. For example, roles were mentioned but not in relation to nurses or the role of nurses was mentioned but not in relation to neuro-advancements. In addition, content related to disabled people was mentioned but not in conjunction with nurses or nursing and in relation to neuro-advancements.

Confirmability, credibility, dependability, and transferability are four trustworthiness measures [95–97]. Confirmability deals with the overall level of confidence that the research study's findings are based on the literature narratives and words rather than potential researcher bias [97]. Using ATLAS.Ti 8<sup>™</sup>, we conducted an audit trail for the coding process by recording data of interest in the literature, recognizing concurrent themes. The authors engaged in peer debriefing on the generation of hit-counts that identified content that should be downloaded (Tables S1–S3 in Supplementary Material) and the manifest and latent coding of the downloaded material. Differences in hit counts from executing our search strategies to obtain articles for qualitative analysis did not occur and neither did differences in manifest coding. Even latent coding differences were few and differences were resolved between the authors. Transferability refers to the extent to which our finding from a study can be applied to other situations. We provided enough data, so others can decide whether to apply our search strategies and analysis to different sources or whether they want to apply different terms such as other neurotechnologies.

#### 2.6. Limitations

Our search was limited to two academic databases and English language literature. As such, our findings are not to be generalized to the whole academic literature, non-academic literature, or non-English literature. Our findings, however, allow for conclusions to be made within the parameters of the searches.

The focus of this study was on Canadian newspapers. Sources such as social media or online only news content (e.g, Canadian Broadcast Corporation) were not the focus and therefore, were not included. Furthermore, the focus was on English-language newspapers only. Therefore, our results cannot be generalized for media in general, newspapers in Canada, and media including newspapers from other countries.

Furthermore, we used 70 neuro-advancement terms (Table 1) based on our prior research. That list is not complete, and one can use more terms if one wants to. Furthermore, new terms might appear in time in the literature as neuro-advancements continue. Therefore, the term list should be updated eventually.

#### 3. Results

The result section was divided into seven parts:

Part One provides the summary of the results from our search strategy that led to the identification of newspaper articles and academic abstracts to be downloaded for content analysis (the full hit count results are in Tables S1–S3 in the Supplemental Material);

Part Two provides hit count for the presence of the 70 neuro-advancement terms (Table 1) in the downloaded academic abstracts;

Part Three covers the latent coding results on the role of nurses;

Part Four covers content related to ethics content and the involvement of nurses in neuroethics and neuro-governance discourses;

Part Five covers lifelong learning;

Part Six covers how disabled people are mentioned;

Part Seven deals with the terms "assistive techolog\*" and "assistive device\*".

All the parts were divided between academic abstracts and newspaper articles.

The results obtained for the academic abstracts and the newspaper were similar. Very few neurotechnologies were mentioned in relation to nurses. The role and skills of nurses mentioned in relation to neuro-advancements focused mostly on clinical practice, if they were mentioned at all, and lifelong learning was rarely engaged with in relation to neuro-advancements and nurses. If lifelong learning was mentioned, the focus was on clinical knowledge or using neurotechnologies (virtual

reality) as an educational tool. Learning about the societal aspects of neuro-advancements through lifelong learning initiatives was not present. Nurses were rarely mentioned in relation to ethics issues and the role of nurses being involved in neuroethics and neuro-governance discussions was absent. Disabled people were mostly engaged with within a medical/therapeutic narrative. Neurotechnologies were rarely identified as assistive technologies or assistive devices and the socio-technical context of assistive technologies in relation to neurotechnologies or in relation to disabled people was absent.

# 3.1. Part 1: Quantitative Results

# 3.1.1. Quantitative Results: Academic

We first generated hit counts for all the search terms. Tables S1 and S2 in the Supplementary Material display the results for the hit counts within academic literature using the two search methods: all search terms within the abstract (Table S1), and "nurs\*" within the source/journal title and neuro-related terms in the abstract (Table S2).

In short, according to Table S1, 43 neuro-related terms had no hits. Eight of these terms had less than ten hits, including the term neurote\* to indicate neurotech related words. The following had more than 10 hits: "neurosc\*" (n = 1944), "Neuroen\*" (n = 434), "virtual reality" (n = 438), "VR" (n = 298), "neuroimaging" (n = 180), "brain stimul\*" (n = 136), "ECoG" (n = 124), "cognitive stim\*" (n = 122), "deep brain stim\*" (n = 120), "DBS" (n = 68), "human computer" (n = 65), "neuromodula\*" (n = 64), "brain imaging" (n = 59), "cochlear implant" (n = 36), "cognitive enhanc\*" (n = 29), "TMS" (n = 26), "transcranial magnetic stim\*" (n = 24), "BCI" (n = 23), "brain computer" (n = 17), "brain-computer interface" (n = 16), "neurostim\*" (n = 15) and "sacral nerve stimulation" (n = 11). Eight were mentioned less than 10 times and 44 were not mentioned at all.

For the search method of "nurs\*" in the source/journal title, and neuro-related terms in the abstract, the following terms had more than 10 hits: neurosc\* (n = 1599), neuroen\* (n = 323), virtual reality (n = 204), brain stimul\* (n = 74), "deep brain stim\*" (n = 71), "VR" (n = 47), "neuroimaging" (n = 36), "ECoG" (n = 29), "human computer" (n = 29), "DBS" (n = 24), "cognitive stim\*" (n = 24), "neuromodula\*" (n = 20), "TMS" (n = 15), "cochlear implant" (n = 15), and "transcranial magnetic stim\*" (n = 14). Fourteen were mentioned less than 10 times and 39 were not mentioned at all.

# 3.1.2. Quantitative Results: Newspaper

As for the newspaper results of search strategy number 2, the full results can be found in Table S3 in the Supplementary Material. In short, the following in conjunction with the term "nurs\*" had more than 10 hits: "neurosc\*" (n = 704), "virtual reality" (n = 343), "VR" (n = 146), "cochlear implant" (n = 137) "brain stim\*" (n = 109), "brain imaging" (n = 103), "DBS" (n = 91), "neuroen\*" (n = 82), "BCI" (n = 78), "deep brain stim\*" (n = 72), "TMS" (n = 51), "transcranial magnetic stim\*" (n = 21) and "neuroimaging" (n = 19). A total of 15 terms had less than 10 hits and 45 had no hits, including "neurote\*".

# 3.2. Part 2: Neuro-Advancements Mentioned in the Downloaded Material

#### 3.2.1. Quantitative Results: Academic

Covering the 1522 abstracts and the 597 abstracts, we found "nurses" or "nursing" mentioned 22 times in conjunction with the term "neuroen\*", of which, the word was in all cases "neuroendocrine" (all within the 1522 abstracts). No hit was obtained for "neurote\*", which could cover the terms "neurotechnology" or "neurotechnologies". The term "neuroscience" was mentioned frequently in conjunction with the terms "nursing" or "nurses" in the 1522 abstracts and rarely in the 597 abstracts, although the mentions were mostly a reflection of the presence of the phrases "neuroscience nurses" and "neuroscience nursing". Within the 1522 abstracts the phrase "neuroscience nurses" was present 184 times and the phrase "neuroscience nursing" was present 476 times, whereby within the 597

abstracts, the phrase "neuroscience nurses" was present three times and the phrase "neuroscience nursing" was not present.

Most of the abstracts (1522 and 597) did not engage with the 70 neuro-related terms (Table 1) in relation to nurses and nursing. The neuro-advancements found were "virtual reality" or "VR " (n = 30), deep brain stimulation or DBS (n = 22), neuroimaging (n = 14), cognitive stimulation (n = 9), cochlear implants, TMS, and transcranial magnetic stimulation (n = 6), spinal cord stimulator and cognitive stimulation (n = 2) and brain computer interface, BCI, cognitive enhancement (n = 1). The rest of the neuro-related terms in Table 1 did not generate any content related to nurses or nursing.

#### 3.2.2. Quantitative Results: Newspaper

We then identified how often neuro-advancements (Table 1) were mentioned in the downloaded newspaper articles in relation to nurs<sup>\*</sup>". Only 6 of the 70 neuro-related terms were found: "virtual reality" (n = 8), cochlear implant<sup>\*</sup> (n = 2), "auditory brainstem<sup>\*</sup>", "transcranial magnetic stim<sup>\*</sup>", "VR", TMS (n = 1).

The newspaper coverage revealed a lack of engagement with nurses in relation to most neurotechnologies.

Our findings reported in Section 3.2 suggest a gap in the coverage of neuro-advancements in relation to nurses and nursing in both the academic literature and the newspapers covered. Most coverage did not engage with specific neuro-advancements. This is problematic because certain applications have specific issues attached to them (see for example, the ethics literature around brain computer interfaces). In addition, the term neurotechnology did not make it into the literature covered. This is a problem given the existence of a neurotechnology governance discourse [11] of relevance to nurses and nursing and that many articles of relevance use the term neurotechnology.

However, hit count does not give the whole story. As such, we reported on the data obtained through a qualitative content analysis in the remainder of the results section.

#### 3.3. Part 3 Qualitative Data: Scope of the Role and Skill of Nursing

#### 3.3.1. Scope of the Role and Skill of Nursing: Academic

Within the 597 abstracts covering the 70 neuro-related terms, the roles associated to nurses in general were occupational-specific and included patient liaisons [98], educators [98–104] and being a leader [99,105]. In turn, the skills included patient assessment [106,107] and patient care [99,108,109]. To reflect such roles and skills in practice, it was outlined that nurses should have the necessary knowledge background in order to practice effectively [98,103,107,110–113]. Outside of the occupational scope, no role of nurses was mentioned.

Within the 597 abstracts, nurses were called to build an increased understanding of different neuro-related issues and the utilization and implementation of various techniques towards appropriate client care, such as appropriate technological knowledge and support [107,109,114].

Some abstracts covering "deep brain stimulation" (DBS) and "brain stimulation" indicated the use and potential of this technology as primarily a tool to aid patients with Parkinson's disease [115–119]. In addition, abstracts covering DBS mentioned the role of patient care and client educators [118,120,121].

Within one abstract covering "spinal cord stim" (SCS), the following role of nurses was mentioned: "caring for patients receiving SCS through patient education, psychologic support, and programming the spinal cord stimulator ... Because of the active role nurses take in pain management, a knowledge of pain transmission and the techniques and efficacy of spinal cord stimulation is important" [109], p. 53. This quote concerns patient education, but the education focuses on clinical aspects, not ethical or social aspects.

The two abstracts covering roles of nurses in relation to "neurostim" both focused on clinical roles and clinical knowledge and therapeutic education [122].

"Transcranial magnetic stim" (TMS), in relation to nurses, indicated operator information [98,105, 123], treatment use [98,123–127], and safety concerns for health professionals [128,129]. One article outlined the role of the nurse in administering repetitive transcranial magnetic stimulation (rTMS) treatment, and the potential for nursing leadership [105]. However, the focus was, again, on the clinical service and not on ethical or governance issues.

Within the 1522 abstracts, if roles and skills of nurses were a topic and linked to neuroadvancements, the vast majority were linked to clinical practice, such as directly related to care services (n = 29). Other roles still linked to clinical practice were advocate for patient/client (n = 5), educator of patient (n = 8), researcher and case manager (n = 5), facilitator and assessor (n = 3), referrer and counsellor and educator of the public on the clinical condition (n = 2), administrator, outreach liaison educator of care givers and staff, enabler of communication of patient, provider for needs of families, advancing ones field, and agent of change [130] (n = 1). To give two quotes:

"A thorough understanding of the latest concepts regarding recognition and management is essential for the neuroscience nurse to play a meaningful role as counselor, educator and clinician in the care of these patients and families". [131], p. 145

"Neuroscience nurses can play an important role in reaching health promotion and disease prevention goals in persons with Parkinson's disease through patient education, advocacy, case management, anticipatory guidance and referrals to primary care providers". [132], p. 229

# 3.3.2. Scope of the Role and Skill of Nursing: Newspaper

Within newspaper literature, we found no article that covered the role of nurses and the scope of nursing in relation to neuro-advancements. We only found roles of nurses in relation to technologies in general. Interestingly, the role of the nurse as a protester/advocate was mentioned twice [133,134], but not in relation to neuro. Most of the skills mentioned were in relation to occupational obligations and general occupational skills [135–139]. Knowledge of technology was mentioned once in relation to the use of computers to help patients with depth perception [140]. Of the mentioned neurotechnologies search terms within the literature, "Virtual Reality", "Cochlear", "Auditory Brainstem" and "Transcranial magnetic stimulation (TMS)", the relation to nurses regarded occupational practice use. "Within these eight articles, it was argued that virtual reality is a useful distraction technique from pain [141,142], as well as that the use of virtual reality to improve nursing education is a good "cost-fighting tool" [143,144]. "Cochlear" in relation to "nurs\*" was mentioned in two articles both covering a nurse in cochlear implantation surgery [145,146]. "Auditory Brainstem" was mentioned once, depicting the process of a nurse conducting an Auditory Brainstem Response Test (ABR) on a patient [147]. Similarly, transcranial magnetic stimulation (TMS) was mentioned once in relation to a nurse utilizing TMS as a treatment method [148].

# 3.4. Part 4: Ethical Implications and Involvement in Neuro-Advancement Governance as a Role of Nurses/Nursing

We found two distinct relationships between nurses and the ethics and governance of neuro-advancements: the first was the ethical implications and knowledge of such ethical issues by the nurses, and the second implication is the role of the nurse in terms of governance and policy discussion surrounding the ethical implications of neuro-advancements. The newspapers only revealed ethical implications and no governance/political involvement of the nurse.

#### 3.4.1. Ethical Implications of Neuro-Advancements for Nurses/Nursing: Academic

Within the 597 abstracts, ethical implications in relation to nurses and neuro-advancements were rarely mentioned. Of the ethical issues mentioned in relation to neuro-advancements, only deep brain stimulation (DBS) n = 2 and Virtual Reality (VR) n = 4 were mentioned. Mentions of DBS indicated the risk of utilizing the technology upon vulnerable patients [149]. Aside from the ethics in

the use of the technology, the ethical implications stated in relation to VR outlined the potential social gaps with the use of this technology in education, particularly outlining the effects on disadvantaged populations [150]. "... major changes associated with VR will occur, and that action must be taken now to guard against the financial discrimination that will affect not only trainees, but also the institutions that treat the least fortunate populations in our society. Medical educators and practitioners should look to our medical leaders to see how we can amplify VR's benefits to patients and trainees while diminishing its adverse effects on medical education and the poorest healthcare institutions" [150], p. 326.

Apart from the ethical issues in the use of neuro-advancements, four abstracts recognized the necessity of academic background and proper ethical training to shape practicing ethics within health care occupations. "Health care trainees rely partly on their training to manage, reason, and reflect on the ethical uncertainties of innovations and new technologies" [149], p. 23. The lack of in-depth coverage of ethics issues and what to do fits with prior findings of our study. One area of ethical contemplation was raised in relation to "cognitive enhance\*", "cognitive stim\*," and the uses of "Cognitive stimulation therapy (CST)" [151–154] with patients and the attitudes of nursing students about the use of cognitive enhancements, where it was found that the decision to use such cognitive enhancers did not occur in isolation and there were many existing social and contextual influences that impacted the decision for nurses to use cognitive enhancers [155].

Within the 1522 neuroscience abstracts, ethics was mentioned in n = 59 abstracts. As for the relevant mentions, one stated: "the role of neuroscience nurses in relation to ethical issues has become increasingly complex. Knowledge of ethical principles and theories assists the nurse in the development of a theoretical basis for resolution of ethical issues or concerns. Additionally, the nurse must have information about practice codes or standards as well as legislative requirements. The nurse must act as an advocate for the patient and society through active participation in institutional ethics committees and legislative forums" [156], p. 621. Others argued that nurses have to be literate on ethical principles and theories [157], for example, because of their involvement in genetic-based neuroscience diseases [158] and because decision-making in nursing practice is impacted by new ways of knowing [159]. One abstract linked neuroscience nurses to ethical decision-making regarding patient advocacy [160], see also [161], and another abstract indicated that the impact of technologies and biological sciences on ethical nursing care remains uncertain [162].

Many of these mentions of ethics could indicate that nurses should be involved in neuroethics and neuro-governance discourses.

# 3.4.2. The Role of Being Involved in Ethics, Policy and Governance Discussion: Academic

Within the 597 abstracts, we found no content that argued for the involvement of nurses or nursing in ethics or neuro-governance discourses or discussions around the social implications of neuro-advancements. Terms such as "decision-making" [163–166], and "leader" [105] were mentioned in relation to clinical aspects and were not linked to ethics or governance discussions. "Expert" and "Responsib\*" indicated the importance of expertise and occupational responsibility in relation to "neuromonitoring, neuroimaging, neurodevelopmental care" [167], p. 421 to better address the needs of their clients, as well as execute techniques [122].

Within the 1522 abstracts, the roles mentioned of nurses did not include having a role in neuro-governance, neuroethics or social implication discussions around neuro-advancements, even if ethical and social implications were acknowledged and the content suggested that nurses should have a role. To give two quote:

"Nursing practice is at a critical ideological and ethical precipice where decision-making is enhanced and burdened by new ways of knowing that include artificial intelligence, algorithms, Big Data, genetics and genomics, neuroscience, and technological innovation". [159], p. 1 Covering traumatic brain injury but no neuro-advancement in particular, one abstract stated that to be a caring nurse includes "reflecting on all sociocultural factors shaping family resources and responses (knowing)" ...; [168], p. E2.

We found two abstracts that indicated that nurses have a role in governance and ethics discourses, although the focus was not on neuro-advancements as such. To give two quotes:

"The role of neuroscience nurses in relation to ethical issues has become increasingly complex ... The nurse must act as an advocate for the patient and society through active participation in institutional ethics committees and legislative forums". [156], p. 621

"The nurse specialist can make a significant contribution within the context of a multidisciplinary team especially in the production of guidelines and policies to ensure and maintain high standards of practice, education for the patient, and the provision of expertise and security that the patient requires when diagnosed with a rare disease". [169], p. 487

3.4.3. Ethical Implications and Governance of Neuro-Advancements Including Role of Nurses/Nursing: Newspapers

We found not one article that engaged with the topic of governance of neuro-advancements in relation to nurses and nursing. Of the ethical implications of neuro-advancements within newspapers, none were linked directly to nurses, however, n = 8 mentions of ethics were placed as considerations within health care. These ethical considerations involved the considerations of cognitive enhancing drugs [170], the public misconception around DBS, "ethicists say patients, often carried away by glowing reports of the procedure's successes, need to be carefully briefed on DBS's unusual side effects, which have included mania, personality changes, marital discord, and increased rates of suicide" [171], p. A8; end-of life decisions for brain-dead patients [172], and technological knowledge enhancement, "how could such knowledge be conveyed through a computer wire? For Hayles, the blurring of these two kinds of knowledge can have serious ethical repercussions" [173], p. F11. Given that no hits were obtained with nurses, the reader will not be triggered to think about nurses as active agents of ethics discussions and being impacted by neuroethics issues.

#### 3.5. Part 5: Lifelong Learning

#### 3.5.1. Lifelong Learning: Academic

As for the 597 abstracts generated from the 70 neuro-related terms (Table 1), the main coverage of education of nurses was linked to the use of "Virtual Reality" (VR) in nursing education (n = 14). The indicated use of VR, however, placed the nurse upon the receiving end of the technology as an educational system/technique, rather than providing education/learning upon the use of virtual reality within the professional setting [100,102,112,163,174–182]. The use of virtual reality simulations was indicated to allow "students to apply and integrate knowledge newly acquired and clinical experience, in order to generate greater security and confidence prior to actual patient arrival" [174], p. 57, as well as to " ... allow trainees to practice techniques without consequences, reduce potential risk associated with training, minimize animal use, and help to develop standards and optimize procedures" [177], p. 67. Only n = 2 abstracts were not related to the use of VR, but rather covered the role of nurses as educators towards educating the patient on deep brain stimulation before and after the intervention [121] and the need for clinical education of nurses for "adequate training both of caregivers and nursing staff in order to assist clinicians in the management of patients in the complicated stages of disease", "utilizing infusional therapies and deep brain stimulation techniques" [183], p. 434. Not one abstract indicated in-depth engagement of operationalizing lifelong learning and none linked lifelong learning as a tool to obtain a more diverse view of how to deal with disabled people beyond the medical view and to be up to date on neuroethical and neuro-governance issues.

Lifelong learning was mentioned in relation to "cochlear implant" in nursing, with the focus on being up to date on clinical knowledge [184–186]. Ethics and governance knowledge or knowledge on

the social discussions around cochlear implants within the deaf community and its implications was not a focus of knowledge.

Some "neuroimaging" abstracts indicated the need to continue nursing education in relation to this technique and provide information towards its use [187–190] but did not engage with knowledge needed to partake in neuroethics and neuro-governance discussions or how to use lifelong learning to obtain the knowledge needed.

As to the 1522 abstracts, education in general was linked to becoming more knowledgeable on the clinical aspects of neuro-advancements and neuro-diseases.

Lifelong learning as a term was not mentioned once. Professional development was mentioned eight times. To give two quotes:

"However, evidence suggests that neuroscience specific education provision is largely unregulated and set outside of a cohesive professional development context" [191], p. 1069 and "Clinical judgment, skills and an evidence-based approach to practice must be maintained through continuing education, training, and the ongoing evaluation of neuro-specific competencies to validate proficiency of care of the neuro rehabilitation patient. These are continuous processes that must be sought by nurses to maintain expertise in the rapidly advancing fields of neuroscience and neurorehabilitation. Professional associations, like the ARN and AANN, provide professional development through education, advocacy, collaboration and research within the specialty". [192], p. 176

Continuing education was mentioned n = 12 times with regard to the role professional nursing organizations have to play [193], continuing education being important for neuroscience nurses [194], and technology supporting continuing education.

"Since its inception in 1968 the American Association of Neuroscience Nurses (AANN) has devoted the majority of its energy and resources to establishing and maintaining excellence in continuing education (CE)". [77], p. 73

# 3.5.2. Lifelong Learning: Newspapers

Within the scope of our newspaper articles, numerous articles covered continuous learning in relation to nurses and nursing, such as "*Canadian Nurses Association launches national online learning network*" [195] and in [196], but there were only five mentions of learning in relation to nurses and our neuro-advancements. The relevant coverage given our focus were in relation to the use of virtual reality and simulation techniques for improving nursing education both for the pre-occupational—nursing students through postsecondary education—as well as the practicing nurse in continuous learning [143,197–199]. As such, learning initiatives was not introduced to readers as a means to learn about the ethical social and other issues linked to neuro-advancements or to indicate that the scope of lifelong learning as a tool was too small.

#### 3.6. Part 6 Narrative of Disabled People

#### 3.6.1. Narrative around Disabled People: Academic

Within the 597 and 1522 abstracts, all the terms that depict disabled people focused on medical and therapeutic aspects in conjunction with the terms. The term patient was the one predominantly used n = 1638. The term "client" was used only 43 times, "disabled people" 13 times and "people with disabilities" 12 times.

When placed in relation to nurses, the roles of nurses were primarily in relation to occupational roles with interacting with the disability community, such as helping to improve patient care [200] and outcomes [201], utilizing neuro-advancement for appropriate patient monitoring [163], the community learning disability epilepsy nurse specialist [202], and mental health nurses [203–205]. The abstracts did not indicate in-depth engagement with the role nurses have in relation to disabled people, especially

the social aspects of the lives of disabled people, and the potential impacts of neuro-linked assistive technologies on such social aspects were missing.

#### 3.6.2. Narrative around Disabled People: Newspapers

Within the newspapers, the narrative around disabled persons was a medical one, whether within the use of the term "patient' or the term "impairment or how the term disability was used. Much of the language surrounding such terms was diagnostic and pertained to medical obligations such as "patient care", "protect patients", etc.

Four articles outlined the potential and use of different types of neurotechnologies by disabled people: deep brain stimulation (DBS) [206], brain control interface (BCI) [207], cochlear implants [106] and virtual reality [208].

Not one article covered disabled people (even under the term patient) in relation to neuroadvancements and nursing. Not one article thematized the relationship between nurses and nursing and disabled people in relation to neuro-advancements, something needed given the contentious debate around the neurotechnology of cochlear implant and certain controversies around brain-computer interfaces.

# 3.7. Part 7 Assistive Technology (AT) as Neuro-Advancements

# 3.7.1. Assistive Technology (AT) as Neuro-Advancements: Academic

Within the 597 abstracts, the term "assistive technolog\*" (AT), obtained n = 8 hit counts, of which n = 6 classified virtual reality as an assistive technology whereby the focus was on therapeutic outcome improvement and prevention of health problems [209–214]. In the studies, the following issues were considered by some researchers in relation to assistive technologies: overall usability [209,213,215], effectiveness [211,212,214], user acceptance [213], and financial feasibility [215]. However, the only role outlined for nurses was in assisting in the use of the technology with patients and no further ethical and social implications were analyzed outside of the medical scope and purpose of such technologies.

There was one abstract that did not explicitly state the term "assistive technology" or "AT", however, the nature of the technology was to modify and further automate wheelchairs for the disability community [215]. According to the authors, the incentive of such a technology was to increase independence and empowerment of the user and will be made to integrate brain computer interfaces (BCI), and Artificial Intelligence (AI) in its modifications to perform functions such as wireless communications and avoiding obstacles [215]. The only relationship to nurses within the article was the risk of potential shortages of medical professionals, such as nurses, with the introduction of such technology.

As for the 1522 abstracts, the term "assistive technolog\*" was only mentioned once in a non-relevant context. However, the term "assistive device\*" was found in one abstract of relevant context that engaged with an eye-tracking device and the Psychosocial Impact of Assistive Devices Scale (PIADS) and the communication of basic needs to nurses [216].

As for the search strategy that focused on the presence of the terms "nurs\*", "assistive technolog\*" and "assistive device\*", in the abstract (search strategy 3, 344 documents), only hearing loss-related devices, which could be seen as a neuro-linked assistive technology, were mentioned three times and one-time virtual reality was mentioned. Only one article covered a role of nurses, namely to help families to cope with hearing loss [217]. The others were not relevant as they compared the cost of a hearing devices with a nursing home [218,219]. VR was mentioned once in relation to nursing homes and not nurses [212].

All remaining abstracts used the terms to assistive devices/assistive technology (AD/AT) without any linkage to neuro-advancements. The role of nurses related to assistive devices/assistive technology (AD/AT) was mainly about nurses using them to make their work lives better and safer (n = 13), nurses involved in providing AT and use of AT on the ward/nursing home to their patients (n = 3) and how

to use AD/AT(n = 2) within their practice. The following themes were mentioned once: "emotional support, information, and access to assistive devices" [220], p. 553, researcher, discharge planning intervention, bad AD used by a nurse, promoting AD to a patient as a home nurse, encountering AD as a home care nurse, AD being needed by nursing home resident, and mandatory inclusion of AT in health professionals curricula [221].

One abstract stated,

"A team of Bioengineering and Nursing faculty delivered an educational and cultural summer experience with the objective to broaden students' knowledge and exposure to health care careers in engineering and nursing, while providing assistive devices and health care outreach to the local community". [222]

# 3.7.2. Assistive Technology (AT) as Neuro-Advancements: Newspaper

Within the newspapers, the term "assistive technolog\*" was only mentioned four times and not once in relation to nursing. This finding indicates that a gap exists between the relationship of nurses with assistive technology and therefore, the relationship between nurses and disabled people, given the impact that various neuro-linked assistive devices have on disabled people.

#### 4. Discussion

Our results indicate similar problems for the academic literature and the newspapers analyzed. Both the academic and newspaper literature covered very few neuro-advancements in relation to nurses and did not, or rarely did, cover governance and engagement, disabled people, assistive technologies, ethics issues and lifelong learning in relation to nurses and neuro-advancements. Finally, if mentioned, the role and skills of nurses mentioned in relation to neuro-advancements focused mostly on clinical practice. We discussed our findings in the remainder of the section through the lens of the role expectations of nurses, the public and stakeholder engagement expectation evident in neuroethics and neuro-governance discourses, the literature dealing with the relationship between nurses and disabled people and the literature engaging with the lifelong learning of nurses.

#### 4.1. The Role of Nurses in Relation to Neuro-Advancements

Nurses are expected to fill many roles and skills [17,18] and all of these roles and skills have the potential to be and therefore, should be applied to neuro-advancements. Our findings suggest that the academic and newspaper coverage around nursing expertise does not extend past the knowledge of using neuro-advancements within clinical services and does not place nurses as a body within the ethical and social implications of neuro-governance and authority of these neuro-advancements. Although the American Nurses Association (ANA) and the American Association of Colleges of Nursing (ANCC) indicate the value of recommending and influencing "social and health care policies" [18] and that the Canadian Nurses Association sees it as important to be "politically astute, by understanding and appropriately intervening in political processes" [17], for all nurses, our findings suggest a gap in the academic and newspaper role narrative of nurses. This gap hinders nurses from adhering to such responsibilities and limits their potential to influence the discussions surrounding neuro-advancements and the dissemination of knowledge of neuro-advancements.

The Canadian Nurses Association pushes for interdisciplinary and collaborative work [11,223,224] and the Framework for the Practice of Registered Nurses in Canada of the Canadian Nurses Association states that

"RNs lead health-care teams, conduct formal research activities, manage nursing services, develop and deliver nursing education, and contribute to the advancement of healthy public policy. They have the skill, expertise and capacity to lead, whether in enhancing client-centred care across the care continuum, directing interprofessional teams or implementing new policy. Nursing leadership is *about critical thinking, action and advocacy—and RNs demonstrate these attributes in all roles and domains of nursing practice".* [224], p. 21

Our findings do not suggest that neuroethics or neuro-governance is part of the interdisciplinary skill and collaborative work scope or that it is a focus of role and skills in relation to policy development, critical thinking, action and advocacy expected for nursing leadership.

Within the academic literature, much of the nurses' interactions with neuro-advancements is within the scope of their clinical practice through treatment administration [122] and proper background knowledge and expertise [107–109,114,122,167]. Furthermore, nurses retain the role of users in relation to neuro-advancements, for example, using virtual reality for educational purposes [100,102,112,163, 174–182].

Roles that would indicate that nurses should and can take part in governance and ethics discourses were absent, indicating a problematic finding.

Clause 8, "Coping effectively with change by proactively envisioning and adapting to a constantly variant healthcare system characterized by rapid change and chaos" and clause 2, "A working knowledge of technology to facilitate the mobility and portability of relationships, interactions, and operational processes'", of the nursing competencies that Huston indicated, outlined that the advancements of technological, societal and occupational aspects are inevitable and thus, nurses must adapt and develop their skills accordingly to match such a rapid ascent [71]. As such, one would have expected a substantial amount of literature, for example, around lifelong learning and how to use it, to be up to date in a continuous fashion on neuroethics and neuro-governance issues.

According to the Canadian Nurses Association, "*it is essential that RNs play an active role in the selection, design, deployment, and evaluation of information and communication technology (ICT) solutions while, at the same time, be given opportunities to acquire ICT competencies to use in their practice*" [224]. Likewise, this same sentiment should be applied beyond information and communication technologies, and into other realms of innovation and technology, such as neuro-advancements. Understanding the implications that neuro-advancements and developing technology pose not only on their professional occupation, but also daily life is important, thus, the push towards nursing engagement in governance decisions and neuro-advancement leadership.

Our findings related to the role narrative are not only problematic given the role expectation of nurses as evident by their own profession, but also in relation to other discourses.

#### 4.2. Ethics: Governance and Public Engagement

Within both the downloaded academic abstracts and the newspaper articles, the neuroethics and ethical coverage of various neuro-advancements is limited in relation to nursing. In addition, the encouragement of nurses to take part in neuroethics and neuro-governance discussions and how to obtain the knowledge needed for such a role was not present at all in the literature. These results also seem to be in sync with other health professionals [225], raising concerns with regards to the willingness and availability of allied health professionals and their voice in the neuro-advancement field.

As indicated earlier, "Health care trainees rely partly on their training to manage, reason and reflect on the ethical uncertainties of innovations and new technologies" [149], p. 23. In addition, The Canadian Nurses Association acknowledges the importance of ethical awareness as "advocating for the discussion of ethical issues among health-care team members, persons receiving care and students, encouraging ethical reflection and working to develop their own and others' awareness of ethics in practice" [226] and the American Nurses Association (ANA) and the American Association of Colleges of Nursing (AACN) recommend that nurses influence social and healthcare policies and apply "legal and ethical standards to complex situations" [18].

As such, there should be more coverage in relation to training nurses in ethical awareness and their involvement in governance discourses.

Health care professionals/practitioners are seen as stakeholders of neuro-advancements [11]. However, our findings show a lack of engagement with the literature covered with the involvement of nurses in the governance of neuro-advancements, despite the ongoing debate within neuroethics and neuro-governance that stakeholders should contribute [11,47,56,227–232].

There must be a communication between all as the governance is one task to be shared by all stakeholders [67]. As was argued by Illes et al., "Indeed, the movement for public understanding is a two-way street where scientists come to understand the public, and science is communicated both in its social context and in socially just ways. This ensures that the requirements for public understanding of both the research process and the process of generating knowledge and scientific evidence are met" [227], p. 8.

Citizen participation is seen to enable citizens to influence policies, programs, and social reforms [233]. However, in order for participation to be influential it must go hand in hand with distribution of power for all groups to benefit [233]. Nurses are also citizens and allowing nurses to take on a broader role within neuro-advancement governance will not only allow for enhanced skill capacity within their roles as professionals but allow for insightful input and opinion from nurses regarding neuro-advancement innovation as professionals and as citizens. Their current positions within their occupation as advocates, researchers, educators, patients support and medical professionals, to name a few, engulf nurses in a variety of interactions and environments, allowing them to understand and see the impact of neuro-advancements from multiple standpoints [17,70,224]. Whether it may be receiving input from patients, teaching students and other health professionals about various neuro-advancements or using these neuro-advancements in their practice, nurses have the potential to act as a useful bridge between the neurofield and the public and other non-clinical settings. Increasing the involvement within the neuroethics and neuro-governance will not only help nurses within their occupation but benefit the neuroethics and neuro-governance discourses and neuro-advancements discourses in general in return, thus enhancing the public and professional collaboration in neuro-advancement governance.

#### 4.3. The Role of the Nurse and Disabled People

Within our findings, the interactions between the nursing field and disabled people were primarily linked to the nurse's occupational obligations, such as appropriate patient care. We did not find further considerations outside of the appropriate occupational interactions with disabled people in our literature. Our findings might not be surprising given that Scullion [234] indicated that disability as a pertinent issue is often neglected in nursing practice, although many nurses hold an obligation to uphold equity and equality in their practice. Although our findings are problematic, they also present an opportunity.

As argued earlier, nurses have the potential to actively take part in and to contribute valuable insight to the discussions and implementation of neuro-advancement governance. In addition, as health care professionals, nurses stand as the mediators between the interactions between disabled people and diverse types of neuro-advancements.

Many studies have analyzed the need for appropriate knowledge of disabled people and different models of disability to be implemented into the nursing practice [234–237]. Thus, with such background knowledge, it is reasonable to enable nurses to advocate for disabled clients and empower disabled clients to advocate for themselves. However, due to their role being so ingrained within the medical field, the conceptions of nurses around disability and the choice of what areas to advocate and empower their clients for can shift between social and medical model influences, which causes a discrepancy in the field. Thus, this poses the question: what perspective should nurses take to appropriately and effectively promote and enable advocacy for, with, and by disabled clients?

In a study conducted by Llewellyn and Northway [238], barriers for nurses towards advocacy were identified and it was found that although all nursing participants were willing to advocate for their clients, the relevance of the advocacy to their field of practice hindered this complete fulfillment [238]. In turn, it was also indicated that

"All nurses recognised barriers to advocacy, but ability to access independent advocacy services was inconsistent, as was participants' attitude towards and willingness to use these. Nurses' received

education in advocacy varied, but they expressed a need for ongoing support and training in advocacy related to their own work area and had specific requirements regarding how and by whom this should be provided". [238]

We submit that working on the social implications of neuro-advancements and to influence neuroethics and neuro-governance discourses is an opportunity for nurses to fulfill the role expectations already mentioned [17,18]. It also an opportunity for nurses to focus on how to engage with the topic of advocacy for, with and by disabled clients, allowing the focus to move beyond the purely medical to include societal, ethical and other aspects. There are concerns and expectations raised related to assistive technologies [80,83] that include neuro-assistive technologies, such as cochlear implants [239–243] and brain computer interfaces [81,244,245], in relation to disabled people and their involvement in the governance and ethics discussions around neuro-advancements is an opportunity to unpack certain issues, such as the identity of disabled people, the perception of disabled people (patient versus consumer) and the consequences of such identity and perception for disabled people. Neuro-assistive devices would be a good case study for revealing certain inconsistencies in the "policy-practice divide" where there is a need for appropriate advocacy and interaction with their clients, however, unfortunately, it is not always necessarily the case. Dealing with the complexities of neuro-assistive technology within ethics and governance discourses is empowering for both groups. It increases the ability of nurses to advocate for and together with disabled people as well as help to enable people with disabilities to take charge and advocate for themselves. Indeed, being part of the governance and ethics discourses will strengthen the advocacy role of nurses.

#### 4.4. Lifelong Learning: Neuro-Advancements

"Not only must individuals acquire initial training as advanced as possible in order to enhance their performance and increase their chances of entering the labour market, but once they have entered the market, they must embark upon a process of lifelong learning to preserve their position in a fiercely competitive environment. The development of the "learning society" further increases the need for individual learning". [74]

Indeed, consistent learning and knowledge provide enhanced skills and abilities that allow employees to retain a competitive edge, however, within healthcare, the necessity for constant learning is the key between life and death [76,174,178]. Nurses are upon the receiving end of neuro-advancements, as seen with virtual reality [100,102,112,163,174–182], as well they are also considered to be the health professionals that should teach, support and administer these various neuro-advancements and innovations to patients [98–104]. Without proper exposure or knowledge, they place themselves and their patients at risk [98,103,107,110–113]. "An RN's comprehensive knowledge base, commitment to lifelong learning and their understanding of clients and the health system, enable them to assume many different roles" [224]. Thus, obtaining the necessary skills and awareness to practice, teach and voice their opinions and concerns is critical within healthcare.

The involvement of developing neuro-governance and neuroethical considerations is critical in guiding informed decision-making and critical thinking within their roles as healthcare professionals *"Health care trainees rely partly on their training to manage, reason and reflect on the ethical uncertainties of innovations and new technologies"* [149], p. 23. Understanding that much of the critical thinking skills, and ethical thought is fostered through educational initiatives indicates the imperative to involve these skills and thoughts into the nursing curriculum. It is suggested that due to the increased potential for the societal impact of neuro-advancements, neuroscience students should be educated in a formal way in neuroethics and public engagement [231]. This same education should be applied to nursing students and professional nurses. In turn, a proper neuro-governance and neuroethics educational background would foster fruitful discussions between nurses and, in turn, neuroscience students about their role within the neuro discourse and the ethical implications they see and face. It might be worthwhile to explore what teaching material is in this sense.

# 5. Conclusions and Future Research

To summarize, our findings indicate that the role narrative of nurses in relation to neuroadvancements were mostly focused on clinical practice, that the role of influencer of neuroethics and neuro-governance discourses was missing, and that lifelong learning did not engage with how to increase the knowledge of nurses on neuroethics and neuro-governance issues to allow them to take part in neuro-related discussions. These findings are problematic for several reasons but also offer opportunities for further research and actions.

It is understood that the societal discussions on science and technology advancements should be a constant endeavor [32,33]. However, our findings highlight the need to better understand why nurses were not engaged within their role of influencers of neuroethics and neuro-governance discourse. It is important to harness this potential and use it properly.

It is also warranted to understand why the impact of neuro-advancements on the relationship between nurses and disabled people was not explored. As much of the nursing clientele are individuals with disabilities, the choices that nurses have in promoting and conducting the use of neuro-advancements and types of neuro-linked assistive technologies for their clients, such as disabled people, affect disabled people and the socio-technical context of the neuro-assistive technology discussions. Thus, fostering effective communication between the opinions of the disability community and the nursing field on such technology could help to promote efficacious yet conscious interventions and provide a needed perspective for the neuro-governance and neuroethics discourses. Nurses are faced with the responsibility of teaching within the workplace to other healthcare associates as well as the public [17,71,114,118,121,224]. Our findings suggest that allowing for their involvement and leadership in neuro-governance and neuroethics engagement can allow for increased professional, client and public awareness of the ethical, legal, and social impacts that neuro-advancements can have upon the public and society. Being part of neuroethics and neuro-governance discussions can be a bridge that brings together nurses and disabled people within a discourse that is of importance to both allowing for an increase in abilities of nurses to engage with disabled people as clients and consumers and to engage and advocate for and with disabled people from a broader knowledge platform then a purely medical one. Our study provides useful data for the nursing, disability, assistive technology, neuroethics and neuro-governance, lifelong learning and policy-related academic literature. It might also be useful for magazines used to inform practitioners and used for advocacy purposes.

It may be fruitful to analyze current nursing professional development and lifelong learning programs to understand what the teaching material is and how it is taught. As for nurses, our study indicates an opportunity to reflect with nurses on our findings related to the role narrative we found and their views on lifelong learning as an instrument to increase the knowledge of nurses on neuroethics and neuro-governance. The further advancement of neuro-assistive devices could be used as a vehicle to better understand the issue of the advocacy of nurses in relation to disabled people and beyond. A recent study outlined future scenarios and trajectories for assisted devices, including neuro-assistive devices [80,82], that could be used to guide that process and to have a discourse between nurses and disabled people in relation to neuro-governance. Our findings suggest that the ethics and governance literature was not pushing for nurses to be part of their discourses, thus indicating another opportunity to better understand neuro-governance realities.

Furthermore, it may be fruitful to consider adapting lifelong learning programs to engage not only with the use and technical background of various neuro-advancements, but also with the ethical, legal, and social implications that these neuro-advancements pose.

**Supplementary Materials:** The following are available online at http://www.mdpi.com/2075-4698/9/3/61/s1, Table S1: Hit count results for articles mentioning the different Neuro-advancements in conjunction with "nurs\*" in the abstracts, Table S2: Hit count results for articles mentioning the different Neuro-advancements in the abstract within journals that had "nurs\*" in the title of the journal, Table S3: Hit count results for Canadian newspaper articles mentioning the different Neuro-advancements in conjunction with the root term "nurs\*".

**Author Contributions:** Conceptualization, R.D. and G.W.; methodology, R.D. and G.W.; formal analysis, R.D. and G.W.; investigation, R.D. and G.W.; data curation R.D. and G.W.; writing—original draft preparation R.D. and G.W.; writing—review and editing, R.D. and G.W.; supervision, G.W.; project administration, G.W.; funding acquisition, G.W.

**Funding:** This research was funded by Government of Canada, Canadian Institutes of Health Research, Institute of Neurosciences, Mental Health and Addiction ERN 155204 in cooperation with ERA-NET NEURON JTC 2017.

Conflicts of Interest: The authors declare no conflict of interest.

# References

- 1. Gross, C.G. Brain, Vision, Memory: Tales in the History of Neuroscience; MIT Press: Boston, MA, USA, 1999.
- 2. Finger, S. Origins of Neuroscience: A History of Explorations into Brain Function; Oxford University Press: Oxford, UK, 2001.
- 3. Glickstein, M. History of Neuroscience; Wiley Online Library: Hoboken, NJ, USA, 2003.
- 4. Moritz, C.T.; Ruther, P.; Goering, S.; Stett, A.; Ball, T.; Burgard, W.; Chudler, E.H.; Rao, R.P. New perspectives on neuroengineering and neurotechnologies: Nsf-dfg workshop report. *IEEE Trans. Biomed. Eng.* **2016**, *63*, 1354–1367. [CrossRef]
- 5. Bassett, D.S.; Khambhati, A.N.; Grafton, S.T. Emerging frontiers of neuroengineering: A network science of brain connectivity. *Annu. Rev. Biomed. Eng.* **2017**, *19*, 327–352. [CrossRef]
- 6. Panuccio, G.; Semprini, M.; Natale, L.; Buccelli, S.; Colombi, I.; Chiappalone, M. Progress in neuroengineering for brain repair: New challenges and open issues. *Brain Neurosci. Adv.* **2018**, *2*, 2398212818776475. [CrossRef]
- Nuffield Council on Bioethics. Novel Neurotechnologies: Intervening in the Brain. Available online: http://nuffieldbioethics.org/wp-content/uploads/2013/06/Novel\_neurotechnologies\_report\_PDF\_web\_0.pdf (accessed on 14 June 2019).
- 8. Department of Neuoroscience, G.U., USA. What is Neuroscience. Available online: https://neuro.georgeto wn.edu/about-neuroscience/ (accessed on 14 June 2019).
- 9. Müller, O.; Rotter, S. Neurotechnology: Current developments and ethical issues. *Front. Syst. Neurosci.* 2017, 11, 93. [CrossRef]
- 10. Garden, H.; Bowman, D.M.; Haesler, S.; Winickoff, D.E. Neurotechnology and society: Strengthening responsible innovation in brain science. *Neuron* **2016**, *92*, 642–646. [CrossRef]
- 11. Garden, H.; Winickoff, D. Issues in neurotechnology governance. Available online: https://www.oecd-ilibrar y.org/content/paper/c3256cc6-en (accessed on 14 June 2019).
- 12. Noroian, E.L.; Yasko, J.M. A survey of neuroscience nurses. J. Neurosurg. Nurs. 1984, 16, 221–227. [CrossRef]
- 13. American Association of Neuroscience Nurses. American association of neuroscience nurses scope of practice statement. *J. Neurosci. Nurs.* **1986**, *18*, 56. [CrossRef]
- 14. Jesudoss, M.; Snegalatha, D.; Ipe, D.M.; Wilson, D.; Bai, R.S. Perception of neuroscience nursing osce among undergraduate nursing students. *Int. J. Nurs. Educ.* **2018**, *10*, 43–47. [CrossRef]
- 15. Weak, L.; Fecske, E.; Breedt, A.; Cartwright, C.; Crawford, D.; Forbes, A. Reflections on 50 years of pediatric neuroscience nursing. *J. Neurosci. Nurs.* **2018**, *50*, 260–264. [CrossRef]
- 16. Canadian Nurses Association. Registered Nurses: Stepping Up to Transform Health Care. Available online: https://www.cnaaiic.ca/~/media/cna/files/en/registered\_nurses\_stepping\_up\_to\_transform\_health\_ care\_e.pdf (accessed on 14 June 2019).
- Canadian Nurses Association. Nursing leadership. Available online: https://www.cna-aiic.ca/-/media/cna/ page-content/pdf-en/nursing-leadership\_position-statement.pdf?la=en&hash=F8CECC6A2D52D8C94EA F939EB3F9D56198EC93C3 (accessed on 14 June 2019).
- Stewart-Amidei, C.; Villanueva, N.; Schwartz, R.; Delemos, C.; West, T.; Tocco, S.; Cartwright, C.; Jones, R.; Blank-Reid, C.; Haymore, J. American association of neuroscience nurses scope and standards of practice for neuroscience advanced practice nurses. *J. Neurosci. Nurs.* 2010, 42, E1–E8. [CrossRef]
- Bell, E.; Leger, P.; Sankar, T.; Racine, E. Deep brain stimulation as clinical innovation: An ethical and organizational framework to sustain deliberations about psychiatric deep brain stimulation. *Neurosurgery* 2016, *79*, 3–10. [CrossRef]
- 20. Glannon, W. Brain, Body, and Mind: Neuroethics with a Human Face; Oxford University Press: Oxford, UK, 2011; pp. 1–272.

- 21. Focquaert, F.; Schermer, M. Moral enhancement: Do means matter morally? *Neuroethics* **2015**, *8*, 139–151. [CrossRef]
- 22. Levin, S.B. Upgrading discussions of cognitive enhancement. Neuroethics 2016, 9, 53-67. [CrossRef]
- 23. Sakakibara, E. Irrationality and pathology of beliefs. Neuroethics 2016, 9, 147–157. [CrossRef]
- 24. Baylis, F. "I am who i am": On the perceived threats to personal identity from deep brain stimulation. *Neuroethics* **2013**, *6*, 513–526. [CrossRef]
- 25. Lipsman, N. What is the role of the neurosurgeon in neuroethical discourse? *Br. J. Neurosurg.* **2013**, 27, 7–8. [CrossRef]
- 26. Rose, S.P.R. 'Smart drugs': Do they work? Are they ethical? Will they be legal? *Nat. Rev. Neurosci.* 2002, *3*, 975–979. [CrossRef]
- 27. Morein-Zamir, S.; Sahakian, B.J. Pharmaceutical Cognitive Enhancement. In *Oxford Handbook of Neuroethics*; Illes, J., Sahakian, B., Eds.; Oxford University Press: Oxford, UK, 2012; pp. 224–244.
- 28. Sahakian, B.J.; Morein-Zamir, S. Neuroethical issues in cognitive enhancement. *J. Psychopharmacol.* **2011**, *25*, 197–204. [CrossRef]
- 29. Cavuoto, J. The market for neurotechnology. J. Med. Mark. 2002, 2, 263–274.
- 30. Wexler, A. The practices of do-it-yourself brain stimulation: Implications for ethical considerations and regulatory proposals. *J. Med. Eth.* **2016**, *42*, 211–215. [CrossRef]
- 31. Human Brain Project (HBP). Human Brain Project. Available online: https://www.humanbrainproject.eu/en/ (accessed on 14 June 2019).
- 32. Cheung, J.; Wolbring, G. Staff's views from one canadian organ procurement organization on organ donation and organ transplant technologies: A content analysis. *Nanoethics* **2017**, *11*, 187–202. [CrossRef]
- 33. Einsiedel, E. Commentary: On the position paper of the ethics committee of the international xenotransplantation association. *Transplantation* **2004**, *78*, 1110–1111. [CrossRef]
- 34. Irwin, A. Sts Perspectives on Scientific Governance. In *The Handbook of Science and Technology Studies*; Hackett, E.J., Amsterdamska, O., Lynch, M., Wajcman, J., Eds.; The MIT Press: Cambridge, MA, USA, 2008; pp. 583–609.
- 35. Turner, J. Democratizing science: A humble proposal. Sci. Technol. Hum. Values 1990, 15, 336–359. [CrossRef]
- 36. Bader, V. Sciences, politics, and associative democracy: Democratizing science and expertizing democracy. *Innovation* **2014**, *27*, 420–441. [CrossRef]
- 37. Kullenberg, C.; Kasperowski, D. What is citizen science? A scientometric meta-analysis. *PLoS ONE* **2016**, *11*, e0147152. [CrossRef]
- 38. Woolley, J.P.; McGowan, M.L.; Teare, H.J.A.; Coathup, V.; Fishman, J.R.; Settersten, R.A., Jr.; Sterckx, S.; Kaye, J.; Juengst, E.T.; Settersten, R.A., Jr. Citizen science or scientific citizenship? Disentangling the uses of public engagement rhetoric in national research initiatives. *BMC Med. Ethics* **2016**, *17*, 33. [CrossRef]
- 39. Goulet, D. Participatory technology assessment: Institution and methods. *Technol. Forecast. Soc. Chang.* **1994**, 45, 47–61. [CrossRef]
- 40. Hennen, L. Participatory technology assessment: A response to technical modernity? *Sci. Public Policy* **1999**, 26, 303–312. [CrossRef]
- 41. Tavella, E. How to make participatory technology assessment in agriculture more "participatory": The case of genetically modified plants. *Technol. Forecast. Soc. Chang.* **2016**, *103*, 119–126. [CrossRef]
- 42. Skorupinski, B. Putting precaution to debate- about the precautionary principle and participatory technology assessment. *J. Agric. Environ. Ethics* **2002**, *15*, 87–102. [CrossRef]
- 43. Durant, J. Participatory technology assessment and the democratic model of the public understanding of science. *Sci. Publ. Policy* **1999**, *26*, 313–319. [CrossRef]
- 44. Guston, D. Understanding 'anticipatory governance'. Soc. Stud. Sci. 2014, 44, 218–242. [CrossRef]
- Diep, L.; Cabibihan, J.-J.; Wolbring, G. Social Robotics Through an Anticipatory Governance Lens. In *Social Robotics*; Beetz, M., Johnston, B., Williams, M.-A., Eds.; Springer International Publishing: New York, NY, USA, 2014; Volume 8755, pp. 115–124.
- 46. Guston, D. The anticipatory governance of emerging technologies. *J. Korean Vac. Soc.* **2010**, *19*, 432–441. [CrossRef]
- 47. Wilsdon, J.; Willis, R. See-Through Science: Why Public Engagement Needs to Move Upstream; Demos: London, UK, 2004.

- 48. Rogers-Hayden, T.; Pidgeon, N. Moving engagement "upstream"? Nanotechnologies and the royal society and royal academy of engineering's inquiry. *Publ. Underst. Sci.* **2007**, *16*, 345–364. [CrossRef]
- 49. Pidgeon, N.; Rogers-Hayden, T. Opening up nanotechnology dialogue with the publics: Risk communication or 'upstream engagement'? *Healthrisk Soc.* **2007**, *9*, 191–210. [CrossRef]
- 50. Krabbenborg, L.; Mulder, H.A.J. Upstream public engagement in nanotechnology: Constraints and opportunities. *Sci. Commun.* 2015, *37*, 452–484. [CrossRef]
- 51. Stilgoe, J.; Owen, R.; Macnaghten, P. Developing a framework for responsible innovation. *Res. Policy* **2013**, 42, 1568–1580. [CrossRef]
- 52. Holitzki, H.; Wolbring, G. Responsible innovation and ethics: A need for globalization. *Eubios J. Asian Int. Bioeth.* **2016**, *26*, 11–18.
- 53. Čeičyte, J.; Petraite, M. The concept of responsible innovation. Publ. Policy Adm. 2014, 13, 400–413. [CrossRef]
- 54. Expert Group on Policy Indicators for Responsible Research and Innovation of the European Commission. Indicators for promoting and monitoring responsible research and innovation. Available online: http://ec.europa.eu/research/swafs/pdf/pub\_rri/rri\_indicators\_final\_version.pdf (accessed on 14 June 2019).
- 55. Grunwald, A. The hermeneutic side of responsible research and innovation. *J. Responsib. Innov.* **2014**, *1*, 274–291. [CrossRef]
- 56. Roskies, A. Neuroethics for the new millenium. Neuron 2002, 35, 21-23. [CrossRef]
- 57. Farah, M.J. Neuroethics: The practical and the philosophical. Trends Cogn. Sci. 2005, 9, 34-40. [CrossRef]
- 58. Levy, N. Introducing neuroethics. *Neuroethics* 2008, 1, 1–8. [CrossRef]
- 59. Soekadar, S.R.; Birbaumer, N. Brain–Machine Interfaces for Communication in Complete Paralysis: Ethical Implications and Challenges. In *Handbook of Neuroethics*; Springer: New York, NY, USA, 2015; pp. 705–724.
- 60. Wallach, W. From robots to techno sapiens: Ethics, law and public policy in the development of robotics and neurotechnologies. *Lawinnov. Technol.* **2011**, *3*, 185–207. [CrossRef]
- 61. Fins, J.J. Neuroethics, neuroimaging, and disorders of consciousness: Promise or peril? *Trans. Am. Clin. Climatol. Assoc.* 2011, 122, 336–346.
- 62. Stahnisch, F.W. History of Neuroscience and Neuroethics: Introduction. In *Handbook of Neuroethics*; Springer: Dordrecht, The Netherlands, 2015; pp. 461–466.
- 63. Di Pietro, N.; Illes, J. Closing gaps: Strength-based approaches to research with aboriginal children with neurodevelopmental disorders. *Neuroethics* **2016**, *9*, 243–252. [CrossRef]
- 64. Nijboer, F.; Clausen, J.; Allison, B.; Haselager, P. The asilomar survey: Stakeholders' opinions on ethical issues related to brain-computer interfacing. *Neuroethics* **2013**, *6*, 541–578. [CrossRef]
- 65. Clausen, J. Conceptual and ethical issues with brain–hardware interfaces. *Curr. Opin. Psychiatry* **2011**, 24, 495–501. [CrossRef]
- 66. Racine, E. Neuroscience, Neuroethics, and the Media. In *Handbook of Neuroethics*; Clausen, J.N.L., Ed.; Springer: Dordrecht, The Netherlands, 2015; pp. 1465–1471.
- Racine, E. Neuroscience and the Media: Ethical Challenges and Opportunities. In Oxford Handbook of Neuroethics; Illes, J., Sahakian, B.J., Eds.; Oxford Press: Oxford, UK, 2012; pp. 783–802.
- 68. Canadian Nurses Association. Registered Nurses, Health and Human Rights. Available online: https://www.cna-aiic.ca/~{}/media/cna/page-content/pdf-en/ps116\_health\_and\_human\_rights\_2011\_e.pdf (accessed on 14 June 2019).
- Canadian Nurses Association. Evidence-Informed Decision-Making and Nursing Practice. Available online: https://www.cna-aiic.ca/-/media/nurseone/page-content/pdf-en/evidence-informed-decision-maki ng-and-nursing-practice.pdf?la=en&hash=D774440B63837990633A0C087F0E12358024318A (accessed on 14 June 2019).
- 70. Canadian Nurses Association. Advanced Nursing Practice. Available online: https://www.cna-aiic.ca/en/nur sing-practice/the-practice-of-nursing/advanced-nursing-practice (accessed on 14 June 2019).
- 71. Huston, C. Preparing nurse leaders for 2020. J. Nurs. Manag. 2008, 16, 905–911. [CrossRef]
- 72. Medel-Anonuevo, C.; Ohsako, T.; Mauch, W. Revisiting Lifelong Learning for the 21st Century. Available online: https://unesdoc.unesco.org/ark:/48223/pf0000127667 (accessed on 14 June 2019).
- 73. OECD. Understanding the Brain: Towards A New Learning Science. Available online: http://www.oecd.org /education/ceri/31706603.pdf (accessed on 14 June 2019).

- 74. OECD's Centre for Educational Research and Innovation (CERI). Learning Sciences and Brain Research. Available online: http://www.oecd.org/education/ceri/brainandlearningprojectbackground.htm (accessed on 14 June 2019).
- 75. Canadian Nurses Association. Promoting Continuing Competence for Registered Nurses. Available online: https://www.cna-aiic.ca/sitecore%20modules/web/~{}/media/cna/page-content/pdf-fr/ps77\_promo ting\_competence\_e.pdf (accessed on 14 June 2019).
- 76. Gopee, N. Professional development. Facilitating the implementation of lifelong learning in nursing. *Br. J. Nurs.* **2005**, *14*, 761–767. [CrossRef]
- Appeldorn, C. Continuing education in neuroscience nursing. J. Neurosci. Nurs. J. Am. Assoc. Neurosci. Nurs. 1988, 20, 73–78. [CrossRef]
- 78. United Nations. Convention on the Rights of Persons with Disabilities (crpd). Available online: https://www. un.org/development/desa/disabilities/convention-on-the-rights-of-persons-with-disabilities.html (accessed on 14 June 2019).
- 79. Goodman, L. Visualisation meets assistive tech: Vr, ar, digital materialisation and the tools for imagining and supporting the full potential of human communication. In Proceedings of the 2016 International Conference on Virtual Systems and Multimedia, VSMM 2016, Kuala Lumpur, Malaysia, 17–21 October 2016; pp. 1–9.
- Nierling, L.; Maia, M.; Bratan, T.; Kukk, P.; Cas, J.; Capari, L.; Krieger-Lamina, J.; Mordini, E.; Wolbring, G. Assistive Technologies for People with Disabilities—Part III: Perspectives, Needs and Opportunities. Available online: http://www.europarl.europa.eu/RegData/etudes/IDAN/2018/603218/EPRS\_IDA(2018)60321 8(ANN3)\_EN.pdf (accessed on 14 June 2019).
- 81. Wolbring, G.; Diep, L. Cognitive/Neuroenhancement Through an Ability Studies Lens. In *Cognitive Enhancement*; Jotterand, F., Dubljevic, V., Eds.; Oxford University Press: Oxford, UK, 2016; pp. 57–75.
- Mordini, E.; Nierling, L.; Wolbring, G.; Maia, M.; Bratan, T.; Capari, L.; Cas, J.; Hennen, L.; Krieger-Lamina, J.; Kukk, P. Assistive Technologies for People with Disabilities—Part II: Current and Emerging Technologies. Available online: https://www.itas.kit.edu/downloads/projekt/projekt\_nier16\_asstech\_EPRS\_IDA\_2018\_6032 18\_ANN2\_EN.PDF (accessed on 14 June 2019).
- 83. Nierling, L.; Maia, M.; Hennen, L.; Wolbring, G.; Bratan, T.; Kukk, P.; Cas, J.; Capri, L.; Krieger-Lamina, J.; Mordini, E. Assistive Technologies for People with Disabilities Part IV: Legal and Socio-Ethical Perspectives. Available online: https://www.itas.kit.edu/downloads/projekt/projekt\_nier16\_asstech\_EPRS\_IDA\_2018\_6032 18\_ANN4\_EN.PDF (accessed on 14 June 2019).
- 84. Grant, M.J.; Booth, A. A typology of reviews: An analysis of 14 review types and associated methodologies. *Health Inf. Libr. J.* **2009**, *26*, 91–108. [CrossRef]
- Davis, K.; Drey, N.; Gould, D. What are scoping studies? A review of the nursing literature. *Int. J. Nurs. Stud.* 2009, 46, 1386–1400. [CrossRef]
- 86. Njelesani, J.; Cameron, D.; Polatajko, H.J. Occupation-for-development: Expanding the boundaries of occupational science into the international development agenda. J. Occup. Sci. 2012, 19, 36–43. [CrossRef]
- Arksey, H.; O'Malley, L. Scoping studies: Towards a methodological framework. *Int. J. Soc. Res. Methodol.* 2005, *8*, 19–32. [CrossRef]
- 88. Anderson, D.; Dumont, S.; Azzaria, L.; Bourdais, M.L.; Noreau, L. Determinants of return to work among spinal cord injury patients: A literature review. *J. Vocat. Rehabil.* **2007**, *27*, 57–68.
- Levac, D.; Colquhoun, H.; O'Brien, K.K. Scoping studies: Advancing the methodology. *Implement. Sci.* 2010, 5, 69. [CrossRef]
- 90. Nord, D.P. A republican literature: A study of magazine reading and readers in late eighteenth-century New York. *Am. Q.* **1988**, *40*, 42–64. [CrossRef]
- 91. Abroms, L.C.; Maibach, E.W. The effectiveness of mass communication to change public behavior. *Annu Rev Publ. Health* **2008**, 29, 219–234. [CrossRef]
- 92. News Media Canada. FAQ. Available online: https://nmc-mic.ca/about-newspapers/faq/ (accessed on 14 June 2019).
- 93. News Media Canada. Snapshot 2016 Daily Newspapers. Available online: https://nmc-mic.ca/wp-content/uploads/2015/02/Snapshot-Fact-Sheet-2016-for-Daily-Newspapers-3.pdf (accessed on 14 June 2019).
- 94. Chai, T.-Y.; Wolbring, G. The portrayal of occupational therapy and occupational science in canadian newspapers: A content analysis. *Societies* **2016**, *6*, 18. [CrossRef]

- 95. Baxter, P.; Jack, S. Qualitative case study methodology: Study design and implementation for novice researchers. *Qual. Rep.* **2008**, *13*, 544–559.
- 96. Lincoln, Y.S.; Guba, E.G. Naturalistic Inquiry; SAGE Publications: Beverly Hills, CA, USA, 1985.
- 97. Shenton, A.K. Strategies for ensuring trustworthiness in qualitative research projects. *Educ. Inf.* **2004**, 22, 63–75. [CrossRef]
- Fitzsimons, L.; Disner, S.G.; Bress, J.N. Effective utilization and future directions for repetitive transcranial magnetic stimulation: A guide for psychiatric nurses. *J. Am. Psychiatr. Nurses Assoc.* 2009, 15, 314–324. [CrossRef]
- 99. Damush, T.M.; Miech, E.J.; Sico, J.J.; Phipps, M.S.; Arling, G.; Ferguson, J.; Austin, C.; Myers, L.; Baye, F.; Luckhurst, C.; et al. Barriers and facilitators to provide quality tia care in the veterans healthcare administration. *Neurology* **2017**, *89*, 2422–2430. [CrossRef]
- Dutile, C.; Wright, N.; Beauchesne, M. Virtual clinical education: Going the full distance in nursing education. Newborn Infant Nurs. Rev. 2011, 11, 43–48. [CrossRef]
- 101. Lahti, M.; Haapaniemi-Kahala, H.; Salminen, L. Use of social media by nurse educator students: An exploratory survey. *Open Nurs. J.* **2017**, *11*, 26–33. [CrossRef]
- 102. Farra, S.L.; Miller, E.T.; Hodgson, E. Virtual reality disaster training: Translation to practice. *Nurse Educ. Pract.* **2015**, *15*, 53–57. [CrossRef]
- 103. Farra, S.; Smith, S.; French, D.; Gillespie, G. Development of an assessment instrument to evaluate performance of the skill of decontamination. *Nurse Educ. Today* **2015**, *35*, 1016–1022. [CrossRef]
- 104. Green, J.; Wyllie, A.; Jackson, D. Virtual worlds: A new frontier for nurse education? *Collegian* **2014**, *21*, 135–141. [CrossRef]
- Bernard, S.; Westman, G.; Dutton, P.R.; Lanocha, K. A psychiatric nurse's perspective: Helping patients undergo repetitive transcranial magnetic stimulation (rtms) for depression. *J. Am. Psychiatric Nurses Assoc.* 2009, 15, 325–332. [CrossRef]
- 106. Engelke, M.K.; Engelke, S.C. Predictors of the home environment of high-risk infants. *J. Community Health Nurs.* **1992**, *9*, 171–181. [CrossRef]
- 107. Ganzer, C.A.; Barnes, A.; Uphold, C.; Jacobs, A.R. Transient ischemic attack and cognitive impairment: A review. *J. Neurosci. Nurs.* **2016**, *48*, 322–327. [CrossRef]
- Halper, J.; Kennedy, P.; Miller, C.M.; Morgante, L.; Namey, M.; Ross, A.P. Rethinking cognitive function in multiple sclerosis: A nursing perspective. *J. Neurosci. Nurs.* 2003, 35, 70–81. [CrossRef]
- Ronk, L.L. Spinal cord stimulation for chronic, nonmalignant pain. Orthop. Nurs. / Natl. Assoc. Orthop. Nurses 1996, 15, 53–58. [CrossRef]
- Lebert, F.; Robert, P.; Rigaud, A.S. Management of behavioral disorders in dementia patients. *Rev. Neurol.* 2000, 156, 767–772.
- 111. Volkmann, J.; Albanese, A.; Antonini, A.; Chaudhuri, K.; Clarke, C.; Bie, R.; Deuschl, G.; Eggert, K.; Houeto, J.-L.; Kulisevsky, J.; et al. Selecting deep brain stimulation or infusion therapies in advanced parkinson's disease: An evidence-based review. *J. Neurol.* **2013**, *260*, 2701–2714. [CrossRef]
- 112. Benham-Hutchins, M.; Lall, M.P. Perception of nursing education uses of second life by graduate nursing students. *CIN Comput. Inform. Nurs.* **2015**, *33*, 404–409. [CrossRef]
- 113. Bakken, S.; Stone, P.W.; Larson, E.L. A nursing informatics research agenda for 2008-18: Contextual influences and key components. *Nurs. Outlook* 2008, *56*, 206–214.e203. [CrossRef]
- 114. O'Brien, A.; Mattison, J.E. Emerging Roles in Health and Healthcare. In *Healthcare Information Management Systems: Cases, Strategies, and Solutions,* 4th ed.; Springer: New York, NY, USA, 2015; pp. 199–217.
- 115. Thevathasan, W.; Gregory, R. Deep brain stimulation for movement disorders. *Pract. Neurol.* **2010**, 10, 16–26. [CrossRef]
- 116. International League Against Epilepsy. Clinical science. Epilepsia (Ser. 4) 2006, 47, 228–232. [CrossRef]
- 117. Okun, M.S.; Fernandez, H.H.; Rodriguez, R.L.; Foote, K.D. Identifying candidates for deep brain stimulation in parkinson's disease: The role of the primary care physician. *Geriatrics* **2007**, *62*, 18–24.
- 118. Lanier-Bohan, E.M.; Heath, S.L. Patient and caregiver perspectives of preoperative teaching for deep brain stimulation surgery. *J. Neurosci. Nurs.* **2016**, *48*, 247–255. [CrossRef]
- Howland, R.H. Update on deep brain stimulation. J. Psychosoc. Nurs. Ment. Health Serv. 2014, 52, 23–26.
   [CrossRef]

- 120. Young, C.; Abercrombie, M.; Beattie, A. How a specialist nurse helps patients undergoing deep brain stimulation. *Prof. Nurse (Lond. Engl.)* **2003**, *18*, 318–321.
- 121. Lanier, E.M.; Buffum, M.D. What are neuroscience nurses teaching parkinson's patients and families before deep brain stimulation? *J. Neurosci. Nurs.* **2011**, *43*, E1–E7. [CrossRef]
- 122. Gayaud, R. Nursing expertise in transcutaneous neurostimulation. Rev. De L'infirmiere 2013, 40-43.
- 123. Anderson, B.S.; George, M.S. A review of studies comparing methods for determining transcranial magnetic stimulation motor threshold: Observation of movement or electromyography assisted. J. Am. Psychiatr. Nurses Assoc. 2009, 15, 304–313. [CrossRef]
- 124. Cheng, C.P.W.; Wong, C.S.M.; Lee, K.K.; Chan, A.P.K.; Yeung, J.W.F.; Chan, W.C. Effects of repetitive transcranial magnetic stimulation on improvement of cognition in elderly patients with cognitive impairment: A systematic review and meta-analysis. *Int. J. Geriatr. Psychiatry* **2018**, *33*, e1–e13. [CrossRef]
- 125. Northoff, G. What is neuroethics? Empirical and theoretical neuroethics. *Curr. Opin. Psychiatry* **2009**, 22, 565–569. [CrossRef]
- 126. Stewart, L.; Walsh, V.; Frith, U.; Rothwell, J.C. Tms produces two dissociable types of speech disruption. *Neuroimage* **2001**, *13*, 472–478. [CrossRef]
- 127. Tredget, J.; Kirov, G.; Dunn, E. Transcranial magnetic stimulation for patients with depression. *Nurs. Stand.* (*R. Coll. Nurs. (G. B. 1987))* 2005, *19*, 49–50. [CrossRef]
- 128. Bottauscio, O.; Zucca, M.; Chiampi, M.; Zilberti, L. Evaluation of the electric field induced in transcranial magnetic stimulation operators. *IEEE Trans. Magn.* **2016**, *52*, 5000204. [CrossRef]
- 129. Zucca, M.; Bottauscio, O.; Chiampi, M.; Zilberti, L. Operator safety and field focality in aluminum shielded transcranial magnetic stimulation. *IEEE Trans. Magn.* **2017**, *53*, 5100704. [CrossRef]
- 130. Murphy, P.; Smith, D.; Vislosky, M.; Tripputi, M.; Cumbler, E.; Blakie, J.; Wilde, M.; Jones, W.; Zachrich, C.; Elser, P.; et al. The impact of a statewide stroke quality improvement program: Colorado stroke alliance and colorado stroke registry. *J. Neurosci. Nurs.* **2011**, *43*, 246–252. [CrossRef]
- 131. Carter, J.H. Tics through tourette's syndrome. J. Neurosci. Nurs. J. Am. Assoc. Neurosci. Nurses 1991, 23, 145–147. [CrossRef]
- 132. Vernon, G.M.; Jenkins, M. Health maintenance behaviors in advanced parkinson's disease. J. Neurosci. Nurs. J. Am. Assoc. Neurosci. Nurses 1995, 27, 229–235. [CrossRef]
- 133. Warick, J. Ruh staff ink warnings: Flight of specialists among concerns cited by angry hospital workers. *Star—Phoenix*, 21 October 2000; A1/FRONT.
- 134. Murray, L. Hospital restructuring plans will destroy years of hard work. *The Ottawa Citizen*, 24 September 1999; F5.
- 135. Alberni Valley Times. Acute care program needs volunteers; helpers provide companionship and interaction to seniors in care at west coast general hospital. *Alberni Valley Times*, 23 October 2014; B11.
- 136. Fantoni, B. Need a neurosurgeon? Expect to wait; extra staff hired to help with backlog. *The Windsor Star*, 15 August 2013; A1.
- 137. Daily Gleaner. Nurses take national certification. Daily Gleaner, 5 July 2000; 1.
- 138. The Gazette. Community calendar. The Gazette, 23 November 1995; H6.
- 139. Other views; editorials from across canada. The Gazette, 18 December 1994; B2.
- 140. Urbanovitch, J. St. Joseph's hospital unveils centre for minimal access surgery. *Stoney Creek News*, 27 October 1999; 26.
- 141. The Vancouver Sun. Learned this week. The Vancouver Sun, 13 October 2007; C6.
- 142. Gruber, B. Virtual reality treatment touted for pain; burn patient goes to play with penguins while agonizing treatment is in progress. *Times—Colonist*, 19 September 2009; D3.
- 143. White, N. Canadas father of cellular joins medsenses to take moncton e-learning firm to next level. *Telegraph—Journal*, 15 March 2007; B1.
- 144. Burgmann, T. Coding on deck for grade-school students as b.C. Unveils new curriculum. *Telegraph—Journal*, 1 February 2016; D3.
- 145. Brown, J. Teen's cochlear implants make 'world of difference'. Toronto Star, 5 May 2011; P12.
- 146. Kelland, K. Pakistani girl shot by taliban to have skull surgery. The Globe and Mail, 31 January 2013; A17.
- 147. Brazelton, T.B. Body function obsession normal in 11-year-old. The Vancouver Sun, 30 July 1999; C11.
- 148. Guardian. Magnets being used to treat variety of mental illnesses. Guardian, 17 June 2003; C1/Front.

- Bell, E.; Racine, E. Clinical and ethical dimensions of an innovative approach for treating mental illness: A qualitative study of health care trainee perspectives on deep brain stimulation. *Can. J. Neurosci. Nurs.* 2013, *35*, 23–32.
- 150. Iserson, K.V. Ethics of virtual reality in medical education and licensure. *Camb. Q Healthc. Eth.* **2018**, 27, 326–332. [CrossRef]
- 151. Apóstolo, J.L.A.; Cardoso, D.F.B.; Rosa, A.I.; Paúl, C. The effect of cognitive stimulation on nursing home elders: A randomized controlled trial. *J. Nurs. Scholarsh.* **2014**, *46*, 157–166. [CrossRef]
- McCabe, L. A holistic approach to caring for people with alzheimer's disease. *Nurs. Stand.* 2008, 22, 50–56.
   [CrossRef]
- 153. O'Reilly, A. Alzheimer's disease. Nurs. Stand. 2009, 23, 59. [CrossRef]
- 154. Yates, L.A.; Yates, J.; Orrell, M.; Spector, A.; Woods, B. *Cognitive Stimulation Therapy for Dementia: History, Evolution and Internationalism*; Routledge: Abingdon, UK, 2017; pp. 1–269.
- 155. Ram, S.; Hussainy, S.; Henning, M.; Stewart, K.; Jensen, M.; Russell, B. Attitudes toward cognitive enhancer use among new zealand tertiary students. *Subst. Use Misuse* **2017**, *52*, 1387–1392. [CrossRef]
- 156. Murphy, W.J.; Olsen, B.J. Ethical perspectives in neuroscience nursing practice. *Nurs. Clin. North Am.* **1999**, 34, 621–635.
- 157. McLeod, A. The sources of ethical deliberation in neuroscience nursing: Three case scenarios. *Br. J. Neurosci. Nurs.* **2005**, *1*, 237–242. [CrossRef]
- 158. Tazbir, J. The human genome project: Ethical and legal considerations for neuroscience nurses. *J. Neurosci. Nurs. J. Am. Assoc. Neurosci. Nurses* **2001**, *33*, 180–183. [CrossRef]
- 159. Peirce, A.G.; Elie, S.; George, A.; Gold, M.; O'Hara, K.; Rose-Facey, W. Knowledge development, technology and questions of nursing ethics. *Nurs. Ethics* **2019**, *26*, 0969733019840752. [CrossRef]
- 160. Fode, N.C. Carotid endarterectomy: Nursing care and controversies. J. Neurosci. Nurs. J. Am. Assoc. Neurosci. Nurses 1990, 22, 25–31. [CrossRef]
- Krueger, D.L. Ethical decision making in neurosurgery: A case study. J. Neurosci. Nurs. 2008, 40, 346–349.
   [CrossRef]
- 162. Peng, Z.Y.; Wan, L.H. Posttraumatic growth of stroke survivors and its correlation with rumination and social support. *J. Neurosci. Nurs.* **2018**, *50*, 252–257. [CrossRef]
- 163. Currie, J.; Bond, R.R.; McCullagh, P.; Black, P.; Finlay, D.D. Vitalsimml—A well-formed data structure to capture patient monitoring scenarios to facilitate the training of nurses via computer-based simulation. In Proceedings of the 2015 Computing in Cardiology Conference (CinC), Nice, France, 6–9 September 2015; pp. 421–424.
- 164. Lee, L.Y.K.; Lee, J.K.L.; Wong, K.F.; Tsang, A.Y.K.; Li, M.K. The establishment of an integrated skills training centre for undergraduate nursing education. *Int. Nurs. Rev.* **2010**, *57*, 359–364. [CrossRef]
- 165. Shamim Khan, M.; Ahmed, K.; Gavazzi, A.; Gohil, R.; Thomas, L.; Poulsen, J.; Ahmed, M.; Jaye, P.; Dasgupta, P. Development and implementation of centralized simulation training: Evaluation of feasibility, acceptability and construct validity. *Bju Int.* 2013, 111, 518–523. [CrossRef]
- 166. Woods, B.; Aguirre, E.; Spector, A.E.; Orrell, M. Cognitive stimulation to improve cognitive functioning in people with dementia. *Cochrane Database Syst. Rev.* **2012**, CD005562. [CrossRef]
- 167. Glass, H.C.; Bonifacio, S.L.; Peloquin, S.; Shimotake, T.; Sehring, S.; Sun, Y.; Sullivan, J.; Rogers, E.; Barkovich, A.J.; Rowitch, D.; et al. Neurocritical care for neonates. *Neurocrit. Care* 2010, 12, 421–429. [CrossRef]
- 168. Roscigno, C.I. Parent perceptions of how nurse encounters can provide caring support for the family in early acute care after children's severe traumatic brain injury. *J. Neurosci. Nurs.* **2016**, *48*, E2–E15. [CrossRef]
- 169. Bouvier, C. Nurse developments in neuroendocrine tumour management. *Endocr. Relat. Cancer* 2003, 10, 487–488. [CrossRef]
- 170. Ogilvie, M. Stronger, faster, smarter?; new research suggests barry bonds' spirit is alive and well among intellectuals. *Toronto Star*, 2 February 2008; ID1.
- 171. Blackwell, T. Beware the miracle cure; parkinson's treatment has worrisome side effects, doctors warn. *National Post*, 23 October 2010; A8.
- 172. DiManno, R. In the post-einstein age, life is more mysterious than ever; where science fails, the heart offers a definition. *Toronto Star*, 21 February 2005; A02.

- 173. Eubanks, V. Taking note of cyborgism: Signs of posthuman life: The universities and multiplex theatres teem with dreams of shedding the body by uploading our minds into machines. *The Vancouver Sun*, 14 August 1999; F11.
- 174. Araya, S.B.; Masalán Apip, M.P.; Paz Cook, M. Health education: In search of innovative methodologies. *Cienc. Y. Enferm.* 2011, 17, 57–69.
- 175. Ball, S.; Bluteau, P.; Clouder, L.; Adefila, A.; Graham, S. Myshoes: An immersive simulation of dementia. In Proceedings of the International Conference on e-Learning, ICEL, Nassau, Bahamas, 25–26 June 2015; pp. 16–23.
- 176. Dubovi, I.; Levy, S.T.; Dagan, E. Now i know how! The learning process of medication administration among nursing students with non-immersive desktop virtual reality simulation. *Comput. Educ.* 2017, 113, 16–27. [CrossRef]
- 177. Engum, S.A.; Jeffries, P.; Fisher, L. Intravenous catheter training system: Computer-based education versus traditional learning methods. *Am. J. Surg.* **2003**, *186*, 67–74. [CrossRef]
- 178. Johannesson, E.; Silén, C.; Kvist, J.; Hult, H. Students' experiences of learning manual clinical skills through simulation. *Adv. Health Sci. Educ.* 2013, *18*, 99–114. [CrossRef]
- 179. Jenson, C.E.; Forsyth, D.M. Virtual reality simulation: Using three-dimensional technology to teach nursing students. *Cin Comput. Inform. Nurs.* 2012, *30*, 312–318. [CrossRef]
- Mantovani, F.; Castelnuovo, G.; Gaggioli, A.; Riva, G. Virtual reality training for health-care professionals. *Cyberpsychol. Behav.* 2003, *6*, 389–395. [CrossRef]
- 181. Fleming, M.; Olsen, D.; Stathes, H.; Boteler, L.; Grossberg, P.; Pfeifer, J.; Schiro, S.; Banning, J.; Skochelak, S. Virtual reality skills training for health care professionals in alcohol screening and brief intervention. *J. Am. Board Fam. Med.* 2009, 22, 387–398. [CrossRef]
- Choi, K.S. Virtual Reality in Nursing: Nasogastric Tube Placement Training Simulator. In MEDINFO 2017: Precision Healthcare Through Informatics; IOS Press: Hangzhou, China; pp. 1298–2018.
- 183. De Rosa, A.; Tessitore, A.; Bilo, L.; Peluso, S.; De Michele, G. Infusion treatments and deep brain stimulation in parkinson's disease: The role of nursing. *Geriatr. Nurs.* **2016**, *37*, 434–439. [CrossRef]
- 184. Stinson, A. Cochlear implantations in children. Aorn J. 1996, 64, 561–562. [CrossRef]
- 185. Goller, Y. Cochlear implantation in children: Implications for the primary care provider. J. Am. Acad. Nurse Pract. 2006, 18, 397–408. [CrossRef]
- Hou, J.H.; Zhao, S.P.; Ning, F.; Rao, S.Q.; Han, D.Y. Postoperative complications in patients with cochlear implants and impacts of nursing intervention. *Acta Oto Laryngol.* 2010, 130, 687–695. [CrossRef]
- 187. Albright, K.C.; Raman, R.; Ernstrom, K.; Hallevi, H.; Martin-Schild, S.; Meyer, B.C.; Meyer, D.M.; Morales, M.M.; Grotta, J.C.; Lyden, P.D.; et al. Can comprehensive stroke centers erase the 'weekend effect'? *Cerebrovasc. Dis.* **2009**, *27*, 107–113. [CrossRef]
- 188. Arif-Rahu, M.; Fisher, D.; Matsuda, Y. Biobehavioral measures for pain in the pediatric patient. *Pain Manag. Nurs.* **2012**, *13*, 157–168. [CrossRef]
- Armstrong, T.S.; Cohen, M.Z.; Weinberg, J.; Gilbert, M.R. Imaging techniques in neuro-oncology. *Semin. Oncol.* Nurs. 2004, 20, 231–239. [CrossRef]
- 190. Jones, B.T.; Kathleen, M. Empirical evidence shows that measuring users' opinions is not a satisfactory way of evaluating computer-assisted learning in nurse education. *Int. J. Nurs. Stud.* **1992**, 29, 411–425. [CrossRef]
- 191. Braine, M.E.; Cook, N. An evaluation of post-registration neuroscience focused education and neuroscience nurses' perceived educational needs. *Nurse Educ. Today* **2015**, *35*, 1069–1074. [CrossRef]
- 192. Aprile, A.; Reilly, K. Nursing Care of the Neuro-Rehabilitation Patient. In *Acquired Brain Injury: An Integrative Neuro-Rehabilitation Approach*; Springer: New York, NY, USA, 2007; pp. 176–187.
- 193. Scahill, L.; Laroche, E.; Bondi, C. Nursing education and clinical practice on child and adolescent psychiatric inpatient services: Survey results. *J. Child Adolesc. Psychiatric Nurs.* **1996**, *9*, 27–34. [CrossRef]
- 194. Hinkle, J.L.; Moran, B.; Prendergast, V.; Reedy, E. Lessons from past presidents of the american association of neuroscience nurses. *J. Neurosci. Nurs.* 2010, 42, E1–E5. [CrossRef]
- Canada NewsWire. Canadian nurses association launches national online learning network. Canada NewsWire, 14 June 2012.
- 196. Krewski, B. Kutz rew. The Oyen Echo, 17 May 2016; A5.
- 197. Metcalfe, Z. The academic neuroscience program; advancing neuroscience care for atlantic canadians neuroscience. *Chronicle—Herald*, 17 June 2016; A9.

- 198. Canada NewsWire. Shriners hospitals for children—Canada opens its pediatric simulation centre. *Canada NewsWire*, 14 September 2016.
- 199. Metcalfe, Z. Bringing life in the or to the small screen. Chronicle—Herald, 25 February 2017; A13.
- 200. Klainin-Yobas, P.; Koh, K.W.L.; Ambhore, A.A.; Chai, P.; Chan, S.W.C.; He, H.G. A study protocol of a randomized controlled trial examining the efficacy of a symptom self-management programme for people with acute myocardial infarction. *J. Adv. Nurs.* 2015, 71, 1299–1309. [CrossRef]
- 201. Schneider, S.M.; Prince-Paul, M.; Allen, M.J.; Silverman, P.; Talaba, D. Virtual reality as a distraction intervention for women receiving chemotherapy. *Oncol. Nurs. Forum* **2004**, *31*, 81–88. [CrossRef]
- 202. Hannah, J.A.; Brodie, M.J. Epilepsy and learning disabilities—A challenge for the next millennium? *Seizure* **1998**, 7, 3–13. [CrossRef]
- 203. Page, S.; Hope, K.; Bee, P.; Burns, A. Nurses making a diagnosis of dementia—A potential change in practice? *Int. J. Geriatr. Psychiatry* **2008**, *23*, 27–33. [CrossRef]
- 204. Horton, L.; Duffy, T.; Hollins Martin, C.; Martin, C.R. Comprehensive assessment of alcohol-related brain damage (arbd): Gap or chasm in the evidence? *J. Psychiatric Ment. Health Nurs.* **2015**, *22*, 3–14. [CrossRef]
- 205. Desocio, J.E. The neurobiology of risk and pre-emptive interventions for anorexia nervosa. *J. Child Adolesc. Psychiatric Nurs.* **2013**, *26*, 16–22. [CrossRef]
- 206. Bradley, L. The tremors are gone. Sudbury Star, 17 June 2005; C1 Front.
- 207. Stonehouse, D. The cyborg evolution: Kevin warwick's experiments with implanting chips to talk to computers isn't as far-fetched as one might think. *The Ottawa Citizen*, 28 March 2002; F2.
- 208. Reynolds, I. Disabled gain virtual mobility; japanese experiments promising. *The Vancouver Sun*, 28 November 2007; A4.
- 209. Chilukoti, N.; Early, K.; Sandhu, S.; Riley-Doucet, C.; Debnath, D. Assistive technology for promoting physical and mental exercise to delay progression of cognitive degeneration in patients with dementia. In Proceedings of the IEEE Biomedical Circuits and Systems Conference Healthcare Technology, BiOCAS2007, Montreal, QC, Canada, 27–30 November 2007; pp. 235–238.
- 210. Burdea, G.; Polistico, K.; Krishnamoorthy, A.; House, G.; Rethage, D.; Hundal, J.; Damiani, F.; Pollack, S. Feasibility study of the brightbrainer<sup>™</sup> integrative cognitive rehabilitation system for elderly with dementia. *Disabil. Rehabil. Assist. Technol.* 2015, 10, 421–432. [CrossRef]
- 211. Dotterer, G. Effects of Assistive Technologies Combined with Desktop Virtual Reality in Instructional Procedures (2). In *Handbook of Research on Human Cognition and Assistive Technology: Design, Accessibility and Transdisciplinary Perspectives*; Medical Information Science Reference: Hershey, PA, USA, 2010; pp. 306–312.
- 212. Grani, F.; Bruun-Pedersen, J.R. Giro: Better biking in virtual reality. In Proceedings of the 2017 IEEE 3rd Workshop on Everyday Virtual Reality, WEVR, Los Angeles, CA, USA, 19 March 2017.
- Kiselev, J.; Haesner, M.; Gövercin, M.; Steinhagen-Thiessen, E. Implementation of a home-based interactive training system for fall prevention: Requirements and challenges. *J. Gerontol. Nurs.* 2015, *41*, 14–19. [CrossRef]
- 214. House, G.P.; Burdea, G.; Polistico, K.; Roll, D.; Kim, J.; Grampurohit, N.; Damiani, F.; Keeler, S.; Hundal, J.; Pollack, S. Integrative rehabilitation of residents chronic post-stroke in skilled nursing facilities: The design and evaluation of the brightarm duo. *Disabil. Rehabil. Assist. Technol.* 2016, 11, 683–694. [CrossRef]
- 215. Mikołajewska, E.; Mikołajewski, D. Selected solutions of automatics and robotics in wheelchairs for disabled people. *Postepy Rehabil.* **2011**, *25*, 11–18.
- 216. Garry, J.; Casey, K.; Cole, T.K.; Regensburg, A.; McElroy, C.; Schneider, E.; Efron, D.; Chi, A. A pilot study of eye-tracking devices in intensive care. *Surgery* **2016**, *159*, 938–944. [CrossRef]
- 217. Landier, W. Hearing loss related to ototoxicity in children with cancer. J. Pediatric Oncol. Nurs. 1998, 15, 195–206. [CrossRef]
- 218. Warland, A.; Tonning, F. Assistive devices for hard-of-hearing persons distribution system and costs. *Br. J. Audiol.* **1991**, 25, 399–404. [CrossRef]
- 219. Jupiter, T. Does hearing assistive technology provide benefit to nursing home residents with dementia? A pilot study. *J. Acad. Rehabil. Audiol.* **2016**, *49*, 34–39.
- 220. Beisecker, A.E.; Cobb, A.K.; Ziegler, D.K. Patients' perspectives of the role of care providers in amyotrophic lateral sclerosis. *Arch. Neurol.* **1988**, *45*, 553–556. [CrossRef]

- 221. Queirós, A.; Alvarelhão, J.; Gonçalves, A.L.; Rocha, N.P. Introducing Assistive Technology in the Formal Education of Healthcare Professionals. In *Assistive Technology Research Series*; IOS Press: Amsterdam, The Netherlands, 2013; Volume 33, pp. 441–445.
- 222. Csavina, K.R.; Zidek, L. A guatemalan immersion for teaching engineering design principles to u.S. High school students. In Proceedings of the ASEE Annual Conference and Exposition, Conference, Vancouver, BC, Canada, 26–29 June 2011.
- 223. Canadian Nurses Association. Interprofessional Collaboration. Available online: https://www.cna-aiic.ca/en /nursing-practice/the-practice-of-nursing/health-human-resources/interprofessional-collaboration (accessed on 14 June 2019).
- 224. Canadian Nurses Association. Framework for the Practice of Registered Nurses in Canada. Available online: https://www.cna-aiic.ca/-/media/cna/page-content/pdf-en/framework-for-the-pracice-of-registere d-nurses-in-canada.pdf?la=en&hash=55716DC66A8C15D13972F9E45BE4AC7AE0461620 (accessed on 14 June 2019).
- 225. Villamil, V.; Wolbring, G. Role and scope coverage of speech-related professionals linked to neuroadvancements within the academic literature and canadian newspapers. *Educ. Sci.* 2019, *9*, 98. [CrossRef]
- 226. Canadian Nurses Association. Code of Ethics for Registered Nurses. Available online: https://www.cna-aiic .ca/en/nursing-practice/nursing-ethics (accessed on 14 June 2019).
- 227. Illes, J.; Blakemore, C.; G Hansson, M.; Hensch, T.; Leshner, A.; Maestre, G.; Magistretti, P.; Quirion, R.; Strata, P. International perspectives on engaging the public in neuroethics. *Nat. Rev. Neurosci.* 2005, *6*, 977–982. [CrossRef]
- 228. Nijboer, F.; Clausen, J.; Allison, B.; Haselager, P. Researchers' opinions about ethically sound dissemination of bci research to the public media. *Int. J. Bioelectromagn.* **2011**, *13*, 108–109.
- 229. Forlini, C.; Partridge, B.; Lucke, J.; Ra, E. Popular Media and Bioethics Scholarship: Sharing Responsibility for Portrayals of Cognitive Enhancement with Prescription Medications; Springer: New York, NY, USA, 2015; pp. 1473–1486.
- 230. Buller, T. Brains, Lies, and Psychological Explanations; Oxford Press: Oxford, UK, 2004; pp. 51-60.
- 231. Morein-Zamir, S.; Sahakian, B.J. Neuroethics and public engagement training needed for neuroscientists. *Trends Cogn. Sci.* 2010, 14, 49–51. [CrossRef]
- 232. Leshner, A.I. It's time to go public with neuroethics. Am. J. Bioeth. 2005, 5, 1–2.
- 233. Arnstein, S.R. A ladder of citizen participation. J. Am. Inst. Plan. 1969, 35, 216–224. [CrossRef]
- 234. Scullion, P. Models of disability: Their influence in nursing and potential role in challenging discrimination. *J. Adv. Nurs.* **2010**, *66*, 697–707. [CrossRef]
- 235. Scullion, P. Education. Enabling disabled people: Responsibilities of nurse education. *Br. J. Nurs.* **2000**, *9*, 1010–1015. [CrossRef]
- 236. Scullion, P. "Disability" in a nursing curriculum. Disabil. Soc. 1999, 14, 539-559. [CrossRef]
- 237. Scullion, P. Conceptualizing disability in nursing: Some evidence from students and their teachers. *J. Adv. Nurs.* **1999**, *29*, 648–657. [CrossRef]
- 238. Llewellyn, P.; Northway, R. An investigation into the advocacy role of the learning disability nurse. *J. Res. Nurs.* **2007**, *12*, 147–161. [CrossRef]
- 239. Cherney, J.L. Deaf culture and the cochlear implant debate: Cyborg politics and the identity of people with disabilities. *Argum. Advocacy* **1999**, *36*, 22–34. [CrossRef]
- 240. Sparrow, R. Defending deaf culture: The case of cochlear implants. J. Political Philos. 2005, 13, 135–152. [CrossRef]
- 241. Blume, S.S. *The Artificial Ear: Cochlear Implants and the Culture of Deafness;* Rutgers University Press: New Brunswick, NJ, USA, 2010.
- 242. Nóbrega, J.D.; de Andrade, A.B.; Pontes, R.J.S.; Bosi, M.L.M.; Machado, M.M.T. Identity of the deaf and interventions in health from the perspective of a community of sign language users. *Cienc. E Saude Coletiva* 2012, 17, 671–679. [CrossRef]
- Sgroi, A.E.; Sinclair, S.J. Cultural implications for mental health professionals working with deaf individuals. *New Sch. Psychol. Bull.* 2018, 15, 55–69.

- 244. Diep, L.; Wolbring, G. Who needs to fit in? Who gets to stand out? Communication technologies including brain-machine interfaces revealed from the perspectives of special education school teachers through an ableism lens. *Educ. Sci.* **2013**, *3*, 30–49. [CrossRef]
- 245. Diep, L.; Wolbring, G. Perceptions of brain-machine interface iechnology among mothers of disabled children. *Disabil. Stud. Q.* **2015**, *35*.



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).